Content-Type: multipart/mixed; boundary="-------------1202170308174"

This is a multi-part message in MIME format.

---------------1202170308174
Content-Type: text/plain; name="12-25.keywords"
Content-Transfer-Encoding: 7bit
Content-Disposition: attachment; filename="12-25.keywords"

NLS, graphs, ground state
---------------1202170308174
Content-Type: application/x-tex; name="saddle13-2-12-bis.tex"
Content-Transfer-Encoding: 7bit
Content-Disposition: inline; filename="saddle13-2-12-bis.tex"

\documentclass[reqno]{amsart}
%\usepackage{amsfonts}
\usepackage{amsmath, amssymb, amsfonts, amsthm}
%\usepackage{epsfig}
%\input{psfig}

%\usepackage[notref,notcite]{showkeys}

%\usepackage[notref,notcite]{showkeys}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}

\setlength{\oddsidemargin}{+5pt}

\setlength{\evensidemargin}{+5pt}

\setlength{\textwidth}{150mm}

\theoremstyle{definition}

\newtheorem{definition}[theorem]{Definition}

\newtheorem{example}[theorem]{Example}

\newtheorem{xca}[theorem]{Exercise}

\newtheorem{cor}[theorem]{Corollary}

\newtheorem{prop}[theorem]{Proposition}

\theoremstyle{remark}

\newtheorem{remark}[theorem]{Remark}
\newcommand{\bes}{{\begin{split}}}
\newcommand{\ees}{{\end{split}}}
\newcommand{\bees}{{\begin{equation}\begin{split}}}
\newcommand{\es}{{\end{split}\end{equation}}}
\newcommand{\erre}{{\mathbb R}}

\newcommand{\enne}{{\mathbb N}}

\newcommand{\natu}{{\mathbb N}}

\newcommand{\comple}{{\mathbb C}}

\newcommand{\inte}{{\mathbb Z}}

\newcommand{\bea}{\begin{eqnarray}}

\newcommand{\eea}{\end{eqnarray}}

\newcommand{\be}{\begin{equation}}

\newcommand{\ee}{\end{equation}}

%\newcommand{\me1}{\left<}

%\newcommand{\me2}{\right<}

\newcommand{\n}{\noindent}

\newcommand{\sob}{{\mathcal L}^{1,2} (L^2 (\erre^n))}

\newcommand{\sobb}{{\mathcal L}^{1,2} (L^2 (\erre^{n+1}))}

\newcommand{\norm}[1]{\left\vert\!\left\vert #1 \right\vert\!\right\vert}

\newcommand{\fine}{\hfill \mbox{$\Box$} \vskip .3cm}
\newcommand{\x}{{\bf{x}}}
\newcommand{\y}{{\bf{y}}}

\newcommand{\ku}{{\bf{k}}}

\newcommand{\F}{{\cal{F}}}

\newcommand{\f}{\frac}

\newcommand{\al}{\alpha}

\newcommand{\la}{\lambda}

\newcommand{\ve}{\varepsilon}

\newcommand{\om}{\omega}

\newcommand{\ov}{\overline}

\newcommand{\U} {\mathcal{U}}

\newcommand{\nnn} {\nonumber}

\newcommand{\tal}{\widetilde \alpha}



\newcommand{\mc}{\mathcal}



\numberwithin{equation}{section}

%    Absolute value notation

\newcommand{\abs}[1]{\lvert#1\rvert}



%    Blank box placeholder for figures (to avoid requiring any

%    particular graphics capabilities for printing this document).

\newcommand{\blankbox}[2]{%

  \parbox{\columnwidth}{\centering

%    Set fboxsep to 0 so that the actual size of the box will match the

%    given measurements more closely.

    \setlength{\fboxsep}{0pt}%

    \fbox{\raisebox{0pt}[#2]{\hspace{#1}}}%

  }%

}

\usepackage{graphicx}
\begin{document}
\begin{thebibliography}{99}
\bibitem{ACFN1}Adami R., Cacciapuoti C., Finco D., Noja D.: Fast solitons on star graphs, Rev Math. Phys. {\bf 23}, 4, 409-451 (2011)
\bibitem{ACFN2}Adami, R. Cacciapuoti C., Finco D., Noja D.: Stationary states of NLS on star graphs arXiv:1104.3839v2 (2011)
\bibitem{BL1} Berestycki H., Lions P.L.: Nonlinear scalar field
  equations I and II, Arch. Rat. Mech. Anal. {\bf 82},   313-375 (1983)
\bibitem{C} Cazenave T.: Semilinear Schr\"odinger equations, AMS (2003)
\bibitem{Ex} Exner P., Keating J.P., Kuchment P., Sunada T., and
  Teplyaev A.: 
Analysis on graphs and its applications, AMS (2008).
\bibitem{Smi}Gnutzmann, S. Smilansky, U., Derevyanko S.:
 Stationary scattering from a nonlinear network, Phys. Rev. A {\bf 83}, 033831 (2011)
\bibitem{Kevr} Kevrekidis P.G., Frantzeskakis D.J., Theocharis G.,
  Kevrekidis I.G.: Guidance of matter waves through Y-junctions,
  Phys. Lett. A {\bf 317}, 51322 (2003)
\bibitem{kostrykin} Kostrykin V., Schrader R.: Kirchhoff's rule for
  quantum wires,  J.Phys. A: Math. Gen. {\bf 32} no. 4, 595-630 (1999)
\bibitem{Kuchment} Kuchment P.: Graph models for waves in thin
  structures, Waves Random Media, {\bf{12}} (4): R1-R24 (2002)
\bibitem{Sob}Sobirov Z., Matrasulov D., Sabirov K., Sawada S.,
  Nakamura K.: 
Integrable nonlinear Schr\"odinger equation on simple networks: 
 Connection formula at vertices, Phys. Rev. E {\bf 81}, 066602 (2010)
\bibitem{TOD}Tokuno A., Oshikawa M., Demler E.: Dynamics of one
  dimesional Bose  liquids: Andreev-like reflection at Y junctions and
  the absence of the  Aharonov-Bohm effect, Phys. Rev. Lett. {\bf 100}, 140402 (2008)
\bibitem{Miro}Miroshnichenko A. E., Molina M.I., Kivshar Y.S.: 
Localized modes and bistable scattering in nonlinear network
junctions,  Phys. Rev. E {\bf 75}, 046602 (2007)
\bibitem{Wan} Wan, W. Muenzel S. Fleischer, W.: Wave Tunneling and
  Hysteresis in Nonlinear  Junctions, Phys. Rev. Lett. {\bf 104}, 073903 (2010)
\end{thebibliography}

\large

\title{On the structure of critical energy levels \\ for the cubic focusing NLS on
  star graphs}





%    Information for first author

\author{Riccardo Adami} 







\address{Adami and Noja: Dipartimento di Matematica e Applicazioni, 
Universit\`a di Milano Bicocca, via R. Cozzi 53, 20125 Milano, Italy
and Istituto di Matematica Applicata e Tecnologie Informatiche ``Enrico Magenes'', CNR, via Ferrata, 1
27100 Pavia, Italy \\
e-mail: riccardo.adami@unimib.it, diego.noja@unimib.it}

\address{
Cacciapuoti: Hausdorff Center for Mathematics, Institut f\"ur
Angewandte Mathematik, 60, Endenicher Allee, 53115 Bonn, Germany \\
e-mail: cacciapuoti@him.uni-bonn.de}

\address{
Finco:  Facolt\`a di Ingegneria, Universit\`a Telematica
Internazionale Uninettuno, 
Corso Vittorio Emanuele II 39, 00186 Roma,
Italy \\
e.mail: d.finco@uninettunouniversity.net}


%``La Sapienza''}





%    \thanks will become a 1st page footnote.





%    Information for second author






\author{Claudio Cacciapuoti}

\author{Domenico Finco}

\author{Diego Noja}


\date{February 11, 2012}


%\dedicatory{This paper is dedicated to our advisors.}



\keywords{}





\begin{abstract} We provide information on a non trivial structure of
  phase space of the cubic NLS on a three-edge star graph.  
We prove that, contrarily to the case of the standard NLS on the line,
the energy associated to the cubic focusing Schr\"odinger equation on
the three-edge star graph with a free (Kirchhoff) vertex does not
attain a minimum value on any sphere of constant $L^2$-norm. We
moreover show that the only stationary state with prescribed
$L^2$-norm is indeed a saddle point.
\end{abstract}


\maketitle

\section{Introduction}

%The definition of nonlinear evolutions on graphs has 
%recently drawn an increasing interest 
%(see e.g. \cite{accn,bellazzini,matrasulov}). 

%and for more information
%on linear dynamics on graphs see \cite{kostrykin,kuchment}).

A major issue in nonlinear dynamics consists in the search for
stationary solutions and in the study of their stability
properties. In the case of hamiltonian systems, a first picture of the
phase space can be drawn by identifying critical points of the energy and their
nature. In particular it is important to know if a {\em ground state}
exists, where a ground state is defined as the minimizer of the energy
functional, possibly restricted to suitable submanifolds. In the
context of the nonlinear Schr\"odinger (NLS) equation, one typically
restricts the energy functional to a manifold on which a second
conserved quantity (sometimes called mass, or charge) is
constant. From a physical point of view, such a restriction is
meaningful, as the extra conserved quantity often represents a
physical characteristic of the system (e.g., the mass, or the number
of particles). In the one dimensional case, the ground states of the
NLS with power nonlinearity on the line are well known and completely
described in the classical paper \cite{BL1}, where more general
nonlinearities are also treated. It turns out that on the line, the NLS
energy constrained to the manifold of the states of constant mass,
attains its minimum value in correspondence of a unique (up to
translation) positive, symmetric and decreasing (for $x>0$)
function. No other critical points of the energy exist. \par\noindent
In this paper we are interested in the case of the focusing nonlinear
cubic Schr\"odinger equation on a three-edge star graph, sometimes called
in the physical literature a Y junction. To put the issue in a
physical context, we begin by recalling that the {\it linear}
Schr\"odinger equation on a graph is a well developed subject, as an
effective description of dynamics of many mesoscopic systems such as,
for example, quantum nanowires (see \cite{Ex, kostrykin, Kuchment} and
references therein). It is of interest to extend the analysis to the
nonlinear wave propagation on networks. In particular, it is well
known that the NLS appears as an effective equation in several
different areas: the description of Bose condensates, the propagation
of electromagnetic pulses in nonlinear (Kerr) media and Langmuir waves
in plasma physics. In many situations it seems of interest to treat
the propagation of NLS solutions associated to such phenomena in one
dimensional ramified structures, the prototype of which is a 
three-edge star graph, or Y-junction. The subject is at its beginnings and
for some preliminary experimental, numerical and analytical works see
\cite{Smi, Kevr, Miro, Sob, TOD, Wan}.  A first rigorous analysis of
nonlinear stationary (or bound) states for NLS on a star graphs is
outlined in \cite{ACFN2}, where solitary waves for a star graph with
delta vertex are constructed, including the case of a free or
Kirchhoff vertex. The Kirchhoff vertex is the closest analogue to the
free particle on the line, to which it reduces when the line is
considered as a graph with two edges. In the linear case the Kirchhoff
laplacian on a graph, in analogy with the laplacian on the line, has
only absolutely continuous spectrum, and so only scattering states are
possible for the Schr\"odinger dynamics. However, in the presence of a nonlinearity a
three-edge star graph with Kirchhoff conditions at the vertex, admits
a unique stationary state, which is quite simply described: it is the
state on the graph which coincides on every edge with half a
soliton. One could suspect that this stationary state is the ground
state. On the contrary, we show that, perhaps unexpectedly, 
this is not the case.



This fact highlights that the NLS on graphs, even on
the simplest one, exhibits remarkable differences
w.r.t. the same evolution equation on the line. 


In this letter we
consider the case of the cubic focusing NLS and analyze the
energy constrained to a fixed sphere in $L^2$, i.e. the set of states
of prescribed mass. We show that the constrained energy is bounded from below (a fact that can be shown similarly
to the case of ${\erre}^n$ and subcritical nonlinearity, see \cite{C} and \cite{ACFN2}), but it approaches the
infimum without attaining a minimum value.
 The more, the only
nonlinear stationary state is a saddle point of the constrained energy
functional. The existence of a saddle point of the energy is a
remarkable feature, inducing on the phase space of the system stable
and unstable manifolds which, in the absence of other critical points,
are the main structural properties of the dynamics. The consequences
of this fact will be investigated in a subsequent paper. \par\noindent
\vskip5pt 
\section{Preliminaries}

Before giving the main results, we start by fixing the
framework, the notation, and recalling some basic results.

\medskip

\n
1. A three-edge star graph ${\mathcal G}$ can be thought of as composed
by  
three halflines with a
common origin, called {\em vertex}. A {\em state} or {\em
  wavefunction} on the graph is an element of the Hilbert space
$L^2 ({\mathcal G}) \ = \ \bigoplus_{i=1}^3 L^2 (\erre^+; dx_i)$, and
can be represented as a
column vector,
namely
$$
\Psi \ = \ \left( \begin{array}{c} \Psi_1 \\
\Psi_2 \\ \Psi_3
\end{array} \right), \qquad \Psi_i \in L^2 (\erre^+).
$$
The space $L^2 ({\mathcal G})$ is naturally endowed with the hermitian
product
$$ ( \Phi, \Psi)_{L^2 ({\mathcal G})} \ = \ \sum_{i=1}^3 \int_0^{+
  \infty} \overline{\Phi_i (x_i)}\Psi_i (x_i) \, dx_i.
$$
Sobolev  
and $L^p$-spaces 
on ${\mathcal G}$ are defined analogously, namely
$$ H^s ({\mathcal G}) \ = \ \ \bigoplus_{i=1}^3 H^s (\erre^+),
\qquad  L^p ({\mathcal G}) \ = \ \ \bigoplus_{i=1}^3 L^p (\erre^+).
$$
In the following, we denote by $\| \Psi \|_p$ the norm of the function
$\Psi$
in the space $L^p ({\mathcal G})$.
When $p=2$ we shall simply write $\| \Psi \|$.

\medskip

\noindent
2. The dynamics of the system is described by the Schr\"odinger
equation
\be \label{schrod}
i \partial_t \Psi (t) \ = \ - \Delta \Psi (t) - |\Psi (t)|^2\Psi (t),
\ee
where:
\begin{itemize}
\item The operator $-\Delta$ acts on the domain
\be \nonumber %\label{domkirch}
D (- \Delta ) \  : = \ \{  \Psi \in H^2 ({\mathcal G}), \ \Psi_1 (0) =
\Psi_2 (0)  =  \Psi_3 (0), \ \Psi^{\prime}_1 (0) +\Psi^{\prime}_2 (0) + \Psi^{\prime}_3 (0)  = 
0 \}
\ee
and its action reads 
$$ - (\Delta \Psi)_i \ = \ - \Psi_i^{\prime \prime}, \qquad i =
1,2,3. $$ The condition at the vertex is usually referred to as the
{\em Kirchhoff's boundary condition}. The operator $- \Delta$ is
selfadjoint on $L^2 ({\mathcal G})$.  Notice that, on the graph, 
the laplacian with
Kirchhoff boundary conditions is the natural generalization 
of the free laplacian on the line, as it is easily seen by considering
the line as a two-edge star graph, and noticing that the
boundary conditions reduces to continuity of the wavefunction and
continuity of the derivative at the vertex, i.e. $\Psi\in H^2(\erre)$.
 
\item The nonlinear term in \eqref{schrod} is defined componentwise, namely
$$ (|\Psi|^2 \Psi)_i : =  |\Psi_i|^2 \Psi_i.$$
\end{itemize}

\medskip

\n
3. The problem \eqref{schrod} is globally well-posed in $H^1
({\mathcal G})$ (see \cite{ACFN1}). The $L^2$-norm and the energy
\be \begin{split} \label{energy}
E (\Psi) & \ = \  \f 1 2 \| \Psi^\prime \|^2 - \f 1 4 \| \Psi \|_4^4 
 \ = \ \sum_{i=1}^3 \left( \f 1 2 \| \Psi_i^\prime \|_{L^2(\erre^+)}^2 - 
\f 1 4 \| \Psi_i \|_{L^4(\erre^+)}^4 \right)
\end{split} \ee
are conserved by time evolution. 

In the sequel we use the following notation:
\be \label{energy1}
 E_1 (\psi)  \ = \  \f 1 2 \| \psi^\prime \|_{L^2(\erre^+)}^2 - \f 1 4
\| \psi \|_{L^4(\erre^+)}^4, \qquad \psi \in H^1 (\erre^+)\ee 
and
\be \label{energy2} E_2 (\psi)  \ = \  \f 1 2 \| \psi^\prime \|_{L^2(\erre)}^2 - \f 1 4
\| \psi \|_{L^4(\erre)}^4,  \qquad \psi \in H^1 (\erre). 
\ee 



\medskip

\n
4.
Let us recall two well-known results on minimization for the cubic
NLS on one- and two-edge star graphs (i.e., on the halfline and on the line):
\begin{enumerate}
\item The minimum of the functional $E_1$ on the functions in $H^1
  (\erre^+)$ with squared $L^2$-norm equal to $m > 0$ is achieved on
  the function (up to a phase factor)
$$ \phi_m (x) \ = \ \f m {\sqrt 2} \cosh^{-1} \left( \f m 2 x \right),
\qquad x \geq 0,$$  
and gives
\be \label{min1}
E_1 (\phi_m ) \ = \ - \f {m^3} {24}. 
\ee
\item 
The minimum of the functional $E_2$ on the functions in $H^1
  (\erre)$ with squared $L^2$-norm equal to $m > 0$ is achieved on
  the functions (up to a phase factor)
\be \label{solit} \varphi_m^y (x) \ = \ \f m {2 \sqrt 2} \cosh^{-1} \left( \f m 4 (x-y)
\right), \qquad x \in \erre 
\ee  
and gives
\be \label{min2}
E_2 (\varphi_m^y ) \ = \ - \f {m^3} {96}. 
\ee
\end{enumerate}

\section{Results}

As recalled in the introduction the energy functional \eqref{energy} at fixed $L^2$ norm is bounded from below for subcritical nonlinearity, which is our case.\par\noindent
Let us introduce the following family of states:
\begin{definition}
We call {\em sesquisoliton} (i.e. ``one and half" soliton) any function of the form
\be \label{sesqui}
\Phi_{m_1, m_2}^x (x_1,x_2,x_3) : = \left(
\begin{array} {c}  \f{m_1}{\sqrt 2} \cosh^{-1} \left(\f{m_1} 2 x_1\right) \\
\f{m_2}{2 \sqrt 2} \cosh^{-1} \left(\f{m_2}{4 } (x_2-x)\right) \\ 
\f{m_2}{2 \sqrt 2} \cosh^{-1} \left(\f{m_2}{4 } (x_3+x)\right)
\end{array} \right)
\ee
where $0 < m_1 \leq m_2$, $x \geq 0$, and the following condition of
``continuity at the vertex'' holds:
\be \label{kirch}
m_1 \ = \ \f {m_2} 2 \cosh^{-1} \left( \f{m_2} 4 x\right).
\ee
\end{definition}

\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.75\textwidth]{sesqui-11feb12.pdf}
\caption{\label{fig1} A sesquisoliton on the three-edge star graph.} 
\end{center}
\end{figure}



\n
Notice that the sesquisolitons are obviously elements of $H^1
({\mathcal G})$. As a matter of fact they also belong to $D (- \Delta
)$.   
\par\noindent
Moreover, in the case $x=0$ one has $m_1=\frac{m_2}{2}$, and one obtains a symmetric configuration with three half solitons concurring at the vertex. 
In \cite{ACFN2} it is shown that this is a standing wave for NLS equation \eqref{schrod}.
\par\noindent
Now, we define the manifold of the sequisolitons with fixed
$L^2$-norm as follows:
\be \nonumber %\label{mani-sesqui}
{\mathcal S}_M : = \{ \Phi_{m_1, m_2}^x, \ \| \Phi_{m_1, m_2}^x \|^2 =
M \}. 
\ee
\begin{theorem} \label{result}
For any $\Psi$ such that $\| \Psi \|^2 = M$, the following 
chain holds:
\be \label{inf}
E (\Psi) > \inf_{\| \Psi \|^2 = M} E (\Psi) = \inf_{\Psi \in {\mathcal
    S}_M} E
(\Psi) = - \f {M^3} {96}.
\ee
\end{theorem}
\begin{proof}
Given $\Psi \in L^2 ({\mathcal G})$, it is possible to construct a
sesquisoliton with the same $L^2$-norm but with lower energy. We
proceed as follows. Let us suppose that 
\be \label{masses} \| \Psi_1 \| \leq \min (\| \Psi_2 \|, \| \Psi_3
\|). \ee
Then, consider the sesquisoliton \eqref{sesqui} with $m_1 =  \|
\Psi_1 \|^2, m_2 = \| \Psi_2 \|^2 + \| \Psi_3 \|^2$, and $x \geq 0$
chosen in order to satisfy the condition \eqref{kirch}. Notice that
such a choice is always possible since $2 m_1 \leq m_2$, and is unique. 

If the condition \eqref{masses} is not fulfilled, then one first
relabels the edges in order to have the minimal mass on the first one,
and thus proceeds as before.

It is immediately
seen that
$ \| \Psi_1 \|^2 = \| (\Phi_{m_1, m_2}^x )_1 \|^2 = m_1$ and 
$ \| \Psi_2 \|^2 +  \| \Psi_3 \|^2 = \| (\Phi_{m_1, m_2}^x )_2 \|^2 +
\| (\Phi_{m_1, m_2}^x )_3 \|^2 = m_2$,
thus $\| \Phi_{m_1, m_2}^x \|_2^2 = M$. 

\n
Let us define the following function on the real line:
\be \label{raddrizza} \begin{split} 
\psi (\xi) \ : = & \ \left\{ \begin{array}{c}
\Psi_2 (- \xi), \qquad \xi < 0 \\
\Psi_3 (\xi), \qquad \xi > 0
\end{array} \right.
\end{split}
\ee
and notice that, by \eqref{solit} and \eqref{sesqui}, 
\be \label{raddrizza2} \begin{split} 
\varphi_{m_2}^{-x} (\xi) \ : = & \ \left\{ \begin{array}{c}
(\Phi_{m_1,m_2}^x)_2 (- \xi), \qquad \xi < 0 \\
(\Phi_{m_1,m_2}^x)_3 (\xi), \qquad \xi > 0
\end{array} \right.
\end{split}
\ee


Furthermore, by \eqref{min1}, \eqref{min2}, \eqref{energy},
\eqref{energy1}, \eqref{energy2}, \eqref{raddrizza}, and \eqref{raddrizza2} one immediately
has the following chain of inequalities
\be \nonumber \begin{split}
E (\Psi) & \ = \ E_1 (\Psi_1) + E_2 (\psi) \ \geq \  E_1 ( \phi_{m_1} )
+  E_2 (\varphi_{m_2}^{-x})
 \ = \ E (\Phi_{m_1, m_2}^x )
%+ E_2 (\phi_{m_2}^x) \\
%& \ = \  E_1 ( \phi
%E_1 ((\Phi_{m_1, m_2}^x )_1) + E_2 (\chi_- (\Phi_{m_1, m_2}^x )_2 +
%\chi_+ (\Phi_{m_1, m_2}^x )_3) \ = \ E_1 (\phi_M) + E_2 (\varphi_M)
\end{split} 
\ee
so the first of the two identities in \eqref{inf} is proven. To prove
the second, we
use \eqref{min1} and \eqref{min2} and obtain
\be \nonumber %\label{e-sesqui}
E (\Phi_{m_1, m_2}^x) \ = \ - \f {m_1^3}{24} - \f {m_2^3}{96}, 
\ee
so, noting that $M = m_1 + m_2$, we obtain
\be \label{e-sesqui-2}
E (\Phi_{m_1, m_2}^x) \ = \  - \f {m_1^3}{24} + \f {(m_1-M)^3}{96},
\ee
where $m_1$ plays the role of a parameter. We stress that, due to the
constraint of the mass of the soliton, $m_1$ can vary in the interval $(0,
M/3]$.
Differentiating \eqref{e-sesqui-2}, one immediately has that $E
(\Phi_{m_1, m_2}^x)$ is monotonically increasing in such interval, so
$$
\inf_{\Psi \in {\mathcal S}_M} E (\Psi) \ = \ \lim_{m_1 \to 0+} E (\Phi_{m_1,
  m_2}^x) \ = \ -  \f {M^3}{96}.
$$
To complete the proof, we must show that, for any $\Psi$ in
$H^1({\mathcal G})$, $E(\Psi)$ is strictly larger than $- M^3/96$.
To this aim, we notice that such an infimum cannot be
achieved, as for $m_1 =0$ the condition \eqref{kirch} does not
correspond to an admissible sesquisoliton, so it cannot be fulfilled.
\end{proof} 

\begin{cor}
The sesquisoliton $\Phi_{M/3, 2M/3}^0$ is a saddle point for the energy
functional.
\end{cor}
\begin{proof}
First, we notice that $\Phi_{M/3, 2M/3}^0$ is a critical point. Indeed,
it satisfies
the Euler-Lagrange equation for the energy functional
constrained on the manifold $\| \Psi \|^2 = M$, namely
$$
- \Delta \Psi - | \Psi |^2 \Psi + \omega \Psi \ = \ 0, 
$$
where $\omega$ is a Lagrange multiplier, coinciding with $\f
{M^2}{36}$.

 
In order to prove that $\Phi_{M/3, 2M/3}^0$ is a saddle point, it is sufficient to show
that it maximizes the energy restricted to a curve to which it belongs in the constraint manifold, and minimizes the energy when restricted to a different curve.

By the proof of theorem
\eqref{result} we have a curve on which $\Phi_{M/3, 2M/3}^0$ is a maximum of the energy, that is, the
curve made of sesquisolitons parametrized by $m_1$. On the other hand,
by \eqref{min1} we know that $\Phi_{M/3, 2M/3}^0$ minimizes the energy
at fixed mass on any edge. Then, it is a minimum of the energy restricted to the submanifold of
function with $\Psi_1 = \Psi_2 = \Psi_3$.
\end{proof}
\vskip 10pt

\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.75\textwidth]{3halfsol-11feb12.pdf}
\caption{\label{fig2}The unique stationary state of the cubic NLS on
  the three-edge star graph.} 
\end{center}
\end{figure}





\n
The previous result can be extended to all star graphs with a similar
construction, and a more systematic analysis of the character of  
stationary states on star graphs will be given in a future work.

\bigskip

\n
{\bf Acknowledgements.} R. A. is partially supported by the PRIN2009 grant 
 ``Critical Point Theory and Perturbative Methods for Nonlinear Differential Equations''.


\vskip20pt

\end{document}



\end{document}

In recent years, the spectacular development of experimental techniques in
condensed matter and ultracold gases, has magnified the interest in
the mathematical modeling of Bose-Einstein condensates, i.e., in the
Gross-Pitaevskii
equation, namely
\begin{equation} \label{gp}
i \partial_t \psi_t (x) \ = \ - \Delta  \psi_t (x) \pm \alpha | \psi_t (x)
|^2 \psi_t (x) + V (x) \psi_t (x), \qquad t \in \erre, \, x \in \erre^d
\end{equation}
where the unknown $\psi_t (x)$ is the wavefunction of the condensate,
$V(x)$ models the action of an external potential, and the cubic term
summarizes the effects of the two-body interactions among the
particles in the condensate. It is well-known that the coupling
constant $\alpha$ is proportional to the scattering length of such a
two-body interaction.  

The function $V$ can be carachterized by the same lengthscale of the
condensate, for instance when it models the trap confining the system,
or by a much shorter lengthscale, for instance when it describes the
effect of an inhomogeneity, or of an impurity. 

In this paper we focus on the latter case, restrict our study to the so-called 
cigar-shaped condensates, i.e., effectively one-dimensional systems, and restrict
to the choice $$V (x) = \delta^\prime_0,$$ 
where, according to the theory of Schr\"odinger operators with
point interactions in dimension one (\cite{albeverio, dombroski}), the symbol $\delta^\prime_0$ does not 
actually refers to a distributional derivative of a Dirac's delta potential, but rather
to the choice of a particular, self-adjoint boundary condition at the origin of the real axis. 
Moreover, we restrict our study to the focusing case (i.e. with the minus sign
in the nonlinearity in \eqref{gp}) and put equal to one the qualitatively irrelevant constant
$\alpha$.

Summarizing all the hypotheses, 
we aim at explicitly finding all stationary solutions to the equation
\begin{equation} \label{problem}
i \partial_t \psi_t (x) \ = \ H \psi_t (x) \pm \alpha | \psi_t (x)
|^2 \psi_t (x), \qquad t \in \erre, \, x \in \erre,
\end{equation}
where $H$ is the so-called ``Schr\"odinger operator with a $\delta^\prime$ potential'', 
i.e., the self-adjoint operator defined on the domain
\be
D (H) \ : = \ \{ \psi \in H^2 (\erre \backslash \{0 \}, 
\ \psi (0+) - \psi (0-) = - \psi^\prime (0+) 
 = - \psi^\prime (0-) \}, 
\ee
whose action reads
\be
(H \psi) (x) \ = \ \psi^{\prime \prime} (x). 
\ee
It is well-known that the standard nonlinear Schr\"odinger equation in dimension one
is an integrable system (\cite{magri}) and that, adding a $\delta$-interaction, 
the stationary states of the system can still be exactly computed (\cite{fukuzumi, fukujean, lacozza}).
In analogy with these two cases, the stationary states for the problem \eqref{problem} prove 
exactly solvable. In addition, as shown in \cite{an2}, the structure of the family of ground
states is here much richer, as it exhibits a sypontaneous symmetry-breaking bifurcation.

Preliminarily, let us fix some notation.

NOTATION


Furthermore, we recall some basic results on well-posedness and
conservation laws

WP and CL 

In the next section we explicitly find the ground states, recall results 
on stability contained in \cite{an2}, and show the appearence of the
bifurcation.  In the last
section we explicitly compute the excited states.



\section{Stationary states}
We call {\em stationary state} any solution to equation
\eqref{problem}
of the type
\be \label{stat}
\psi_\om (t,x) \ = \ e^{i \om t} \phi_\om (x).
\ee
As a consequence, the square-integrable
function $\phi_\om$ must solve the {\em
  stationary Schr\"odinger equation}
\be \label{stat-eq}
H \phi_\om - | \phi_\om |^2  \phi_\om + \om \phi_\om \ = \ 0.
\ee
Equation \eqref{stat-eq} can be rephrased as
\be \label{stat-eq} \begin{split} &
- \phi_\om^{\prime \prime}
 - | \phi_\om |^2  \phi_\om + \om \phi_\om \ = \ 0, \qquad x \neq 0 \\
& \phi_\om (0+) - \phi_\om (0-) \ = \ - \phi_\om^\prime (0+)  \ = \ -
\phi_\om^\prime (0-). 
\end{split}
\ee

In the following we reduce equation
\eqref{stat-eq} to a couple of systems of two equations in two real
unknowns. For the convenience of the reader, we preliminary solve such
systems.
\begin{prop}
The solutions $(t_1, t_2)$ to the system
\be 
\label{tipiu}
T_+ : = \left\{  \begin{array}{ccc}
t_1^2 - t_1^4 & = & t_2^2 - t_2^ 4
\\ & & \\
t_1^{-1} - t_2^{-1} & = & \sqrt \om
\end{array} \right.
\ee
with the condition $0 \leq t_1, t_2 \leq 1$, are given by
\be \begin{split} \label{solt+}
t_1 \ = & \ \f{1 - \sqrt{1 + \om} + \sqrt{ \om - 2 + 2 \sqrt{1 + \om}}}
{2 \sqrt \om} \\
t_2  \ = & \ \f{- 1 + \sqrt{1 + \om} + \sqrt{ \om - 2 + 2 \sqrt{1 + \om}}}
{2 \sqrt \om}
\end{split} \ee
Such solutions exist for any $\om > 0$.
\end{prop}
\begin{proof}
the first equation in \eqref{tipiu} rewrites as
$$
(t_1^2 - t_2^2) (t_1^2 + t_2^2 - 1) \ = \ 0,
$$
so the system splits into two subsystems $T_{1,+}$ and $T_{2,+}$, defined by
\be T_{1,+} : = \left\{  \begin{array}{ccc}
t_1 & = & t_2
\\ & & \\
t_1^{-1} - t_2^{-1} & = & \sqrt \om
\end{array}
\right. \nonumber
\ee
and
\be T_{2,+} : = \left\{  \begin{array}{ccc}
t_1^2 + t_2^2 & = & 1
\\ & & \\
t_1^{-1} - t_2^{-1} & = & \sqrt \om
\end{array}
\right. \nonumber
\ee
The system $T_{1,+}$ can be solved only for $\om = 0$, so we are not
interested  in such a solution

Let us consider the system $T_{2,+}$. Multiplying the second equation by $t_1t_2$ 
one obtains
\be \label{summa}
t_1 - t_2 \ = \ \sqrt \om t_1 t_2,
\ee
thus squaring both sides, using the first equation of $T_2$ and
imposing positivity,   we obtain
$$
t_1 t_2 \ = \ \f{\sqrt{1 +  \om} - 1} {\om}
$$
and so, by \eqref{summa}
\be \label{strauss}
t_2 - t_1 \ = \ \f{\sqrt{1 +  \om} - 1}{\sqrt \om}.
\ee
By \eqref{strauss} and the first equation of $T_{2,+}$, we finally get
\eqref{solt+}.
%
%\be \label{kahn}
%t_{1,2} \ = \ \f{ \beta \pm \sqrt{2 - \beta^2}} 2
%\ee
%with $\beta : =  \f{1 + \sqrt{1 + \gamma^2 \om}}{\gamma \sqrt
%  \om}$. The positivity of the quantity under square yields
%the condition $\om > \om^*$. By \eqref{kahn} the proof of 2. is complete.
\end{proof}



\begin{prop}
The solutions $(t_1, t_2)$ to the system
\be 
\label{timeno}
T_- : = \left\{  \begin{array}{ccc}
t_1^2 - t_1^4 & = & t_2^2 - t_2^ 4
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \sqrt \om
\end{array} \right. \ee
with the condition $0 \leq t_1, t_2 \leq 1$, can be classified as
follows: 
\begin{enumerate}
\item For $0 < \om \leq 4$ there are no solutions.
\item For $\om > 4$ there esists the solution
\be \label{t-symm}
t_1 \ = \ t_2 \ = \ \f 2 {\sqrt \om}.
\ee
\item For $\om > 8$ there exist the two solutions
\be \begin{split} \label{solt-asym}
t_1 \ = & \ \f{1 + \sqrt{1 + \om} \pm \sqrt{ \om - 2 - 2 \sqrt{1 + \om}}}
{2 \sqrt \om} \\
t_2  \ = & \ \f{ 1 + \sqrt{1 + \om} \mp \sqrt{ \om - 2 - 2 \sqrt{1 + \om}}}
{2 \sqrt \om}
\end{split}
\ee
\end{enumerate}
\end{prop}
\begin{proof}
As in the previous proof, we rewrite the first equation in
\eqref{solt-asym} as
$$
(t_1^2 - t_2^2) (t_1^2 + t_2^2 - 1) \ = \ 0,
$$
so the system \eqref{solt-asym} is equivalent to the union of 
the systems
\be T_{1,-} : = \left\{  \begin{array}{ccc}
t_1 & = & t_2
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \sqrt \om
\end{array}
\right. \nonumber
\ee
and
\be T_{2,-} : = \left\{  \begin{array}{ccc}
t_1^2 + t_2^2 & = & 1
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \sqrt \om
\end{array}
\right. \nonumber
\ee
The system ${T_{1,-}}$ gives the solution \eqref{t-symm}. The condition $\om >
4$ emerges by the prescription $t_1, t_2 < 1$, so point {\em 1.} is
proven. 

Let us consider the system $T_{2,-}$. Multiplying the second equation by $t_1t_2$ 
one obtains
\be \label{summa2}
t_1 + t_2 \ = \  \sqrt \om t_1 t_2,
\ee
thus squaring both sides, using the first equation of $T_{2,-}$ and
imposing positivity,   we have
$$
t_1 t_2 \ = \ \f{1 + \sqrt{1 +  \om}} { \om}
$$
and so, by \eqref{summa2}
\be \label{strauss2}
t_1 + t_2 \ = \ \f{1 + \sqrt{1 +  \om}}{ \sqrt \om}.
\ee
By \eqref{strauss2} and the first equation of ${T_{2,-}}$, we finally get
\eqref{solt-asym}. The positivity of the quantity under the square
root of \eqref{solt-asym} gives the condition $\om > 8$, so the proof
is complete.
\end{proof}


Now we can explicitly give all the solutions to \eqref{eq-stat}.

\begin{theorem}\label{statstat}
The stationary states of equation \eqref{problem} can be classified as
follows:
\begin{enumerate}
\item The unperturbed symmetric solitary wave
\be \label{unperturbed} 
\phi_{\om}^{0,0} \ : = \ 
\sqrt{2 \om} \cosh^{-1} (\sqrt\om x).
\ee
Such solutions are present for any $\om > 0$.

\item The asymmetric non-changing sign solutions
%, as either $\phi_{\om,+}^{x_1, x_2}$ or
%  $\phi_{\om,-}^{x_1, x_2}$, where
\be \label{stazzionario2} 
\phi_{\om, \pm}^{x_1, x_2} \ : = \ 
\begin{array}{ccc}
\sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_1)), & & x< 0 \\
\sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_2)), & & x< 0,
\end{array}
\ee
\be \begin{split}
x_1 \ = & \ \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}, \quad
x_2 \ =  \ \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2} \\ \\
x_1 \ = & \ - \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2}, \quad 
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}
\end{split} \ee,
where the couple $(t_1, t_2)$ solves the system \eqref{t1}.

\noindent
Such solutions are present for any $\om > 0$.

\item  The changing sign solutions, which, in turn, can be
classified as
\begin{enumerate}
\item the antisymmetric solutions
\be \label{stazzionario} 
\phi_{\om, -}^{\bar x, - \bar x} \ : = \ 
\begin{array}{ccc}
\sqrt{2 \om} \cosh^{-1} (\sqrt\om (x- \bar x)), & & x< 0 \\
- \sqrt{2 \om} \cosh^{-1} (\sqrt\om (x + \bar x)), & & x > 0,
\end{array}
\ee
where 
\be \label{xbar}
\bar x = \f 1 {2 \sqrt \om} \log \f{\sqrt \om + 2} {\sqrt \om - 2}.
\ee
Such solutions are present for $\om > 4$.


\item
\be \label{stazzionario2} 
\phi_{\om, \pm}^{x_1, x_2} \ : = \ 
\begin{array}{ccc}
\sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_1)), & & x< 0 \\
- \sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_2)), & & x< 0,
\end{array}
\ee

\be \begin{split}
x_1 \ = & \ \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}, \quad
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2} \\
x_1 \ = & \  \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2}, \quad
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}
\end{split} \ee,
where the couple $(t_1, t_2)$ solves the system \eqref{timeno}

\noindent
Such solutions are present for any $\om > 8$.
\end{enumerate}
\end{enumerate}
\end{theorem}


\begin{proof}
Equation \eqref{eqstat} can be separately solved in the positive and in the
negative halfline. It is well-known that the whole family of
$L^2$-solutions is given by
\be 
\phi_\om^{x_1, x_2, \theta_1, \theta_2} \ : = \ 
\begin{array}{ccc}
e^{i \theta_1} \sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_1)), & & x< 0 \\
e^{i \theta_2} \sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_2)), & & x< 0.
\end{array}
\ee
Since both equations \eqref{problem}, \eqref{stat-eq} are invariant
under multiplication by a phase factor, one can always fix $\theta_1 =
0$. Furthermore, the matching conditions at the origin can hold only
if either $\theta_2 = 0$ or $\theta_2 = \pi$, so the family of
solution to \eqref{eq-stat} simplifies to
\be \label{stazzionario} 
\phi_{\om, \pm}^{x_1, x_2} \ : = \ 
\begin{array}{ccc}
\sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_1)), & & x< 0 \\
\pm \sqrt{2 \om} \cosh^{-1} (\sqrt\om (x-x_2)), & & x< 0.
\end{array}
\ee
To fix parameters $x_1$ and $x_2$ one imposes the matching
conditions, and obtains that the
couples $(x_1, x_2)$ are to be determined as follows:

\noindent
-- For the family of changing-sign solutions $\phi_{\om,+}^{x_1,
  x_2}$, 
one solves the
system
\be T_+ : = \left\{  \begin{array}{ccc}
t_1^2 - t_1^4 & = & t_2^2 - t_2^ 4
\\ & & \\
t_1^{-1} - t_2^{-1} & = & \sqrt \om
\end{array}
\right. ,  \qquad 0 \leq t_1, t_2 \leq 1. \nonumber
\ee
Then, the possible couples $(x_1, x_2)$ are two, and they
are determined by
\be \begin{split}
x_1 \ = & \ \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}, \quad 
x_2 \ =  \ \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2} \\
x_1 \ = & \ - \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2}, \quad 
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}
\end{split} \ee

\noindent
-- For the family of changing sign solutions $\phi_{\om,-}^{x_1,
  x_2}$, 
one solves the
system
\be T_- : = \left\{  \begin{array}{ccc}
t_1^2 - t_1^4 & = & t_2^2 - t_2^ 4
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \sqrt \om
\end{array}
\right. ,  \qquad 0 \leq t_1, t_2 \leq 1. \nonumber
\ee
Then, the possible couples $(x_1, x_2)$ are two, and they
are determined by
\be \begin{split}
x_1 \ = & \ \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}, \quad
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2} \\
x_1 \ = & \ \f 1 {2 \sqrt \om} \log \f{1+ t_2} {1 - t_2}, \quad 
x_2 \ =  \ - \f 1 {2 \sqrt \om} \log \f{1+ t_1} {1 - t_1}
\end{split} \ee
\end{proof}

By a direct comptutation, one can prove the following
\begin{proposition}
The $L^2$-norm, the $L^4$-norm, and the energy of the stationary
states found in Theorem \ref{statstat}, are explicitly given by
\begin{eqnarray}
\| \psi_{\om,+}^{0,0} \|_2^2 \ = \ 4 \sqrt \om & \| \psi_{\om,+}^{0,0}
\|_4^4 \ = \ \f {16} 3 \om^{\f 3 2} & E ( \psi_{\om,+}^{0,0} ) \ = \
- \f 2 3 \om^{\f 3 2} \\
S_\om ( \psi_{\om,+}^{0,0} ) \ = \
 \f 4 3 \om^{\f 3 2} & & \\
\| \psi_{\om,+}^{x_1,x_2} \|_2^2 \ = \ 2 \sqrt \om \left( 2 + t_2 -
  t_1
\right) & & \| \psi_{\om,+}^{x_1,x_2} \|_4^4 \ = \ 4 \om^{\f 3 2}
\left[ \f 4 3 + (t_2 - t_1) - \f 1 3 (t_2^3 - t_1^3) \\
E (\psi_{\om,+}^{x_1,x_2}) \ = \ - \f 2 3 \om^{\f 3 2} + \f {2 - \om}
3 (\sqrt{\om + 1} - 2 ) & &
S_\om (\psi_{\om,+}^{x_1,x_2}) \ = \  \f 4 3 \om^{\f 3 2} + \f 2 3
(\om + 1)^{\f 3 2} - \om - \f 2 3 \\
\| \psi_{\om,-}^{\bar x, - \bar x} \|_2^2 \ = \ 4 \sqrt \om - 8 &
\| \psi_{\om,+}^{\bar x, - \bar x} \|_4^4 \ = \ \f {16} 3 \om^{\f 3 2} - 16 \om
+ \f {64} 3 \\
E (\psi_{\om,+}^{\bar x, - \bar x}) \ = \ \f {16} 3 - \f 2 3 \om^{\f 3 2} &
S_\om (\psi_{\om,+}^{\bar x, - \bar x}) \ = \ \f {16} 3 - 4 \om + \f 4 3
\om^{\f 3 2} + \f 2 3 (\om + 1)^{\f 3 2} - \om - \f 2 3 \\
\| \psi_{\om,-}^{x_1,x_2} \|_2^2 \ = \ 2 \sqrt \om \left( 2 + t_2 -
  t_1
\right) & & \| \psi_{\om,+}^{x_1,x_2} \|_4^4 \ = \ 4 \om^{\f 3 2}
\left[ \f 4 3 + (t_2 - t_1) - \f 1 3 (t_2^3 - t_1^3) \\
E (\psi_{\om,+}^{x_1,x_2}) \ = \ - \f 2 3 \om^{\f 3 2} + \f {2 - \om}
3 (\sqrt{\om + 1} - 2 ) & &
S_\om (\psi_{\om,+}^{x_1,x_2}) \ = \  \f 4 3 \om^{\f 3 2} + \f 2 3
(\om + 1)^{\f 3 2} - \om - \f 2 3 \\

\section{Energy of the stationary states}

\section{Bifurcations}




ere we specialize to the case with $\mu = 1$, i.e., the cubic
nonlinearity. As widely known, the unperturbed problem gives an
integrable system. The presence of a defect prevents the system from
being fully integrable, nevertheless it remains possible to perform
exact computations. The results can be summarized as follows
\begin{theorem} \label{cubicat}
Fix $\mu = 1$. Then, the solution to the system \eqref{tsystem} are
\begin{enumerate}
\item The symmetric one $t_1 = t_2 = \f 2 {\gamma \sqrt \om}$, for
  any $\om > \om_0$.
\item The asymmetric one
\be \label{cubicasimm} \begin{split}
t_1 \ : = \ & \f {1 + \sqrt{1 + \gamma^2 \om} - \sqrt{\gamma^2 \om - 2 - 2
    \sqrt{1 + \gamma^2 \om}}} {2 \gamma^2 \om} \\
t_2 \ : = \ & \f {1 + \sqrt{1 + \gamma^2 \om} + \sqrt{\gamma^2 \om - 2 - 2
    \sqrt{1 + \gamma^2 \om}}} {2 \gamma^2 \om},
\end{split}
\ee
and the solution obtained exchanging $t_1$ and $t_2$, for any $\om > \om^*$.
\end{enumerate}
\end{theorem}

\begin{proof}
Fixed $\mu = 1$, the first equation in \eqref{tsystem} rewrites as
$$
(t_1^2 - t_2^2) (t_1^2 + t_2^2 - 1) \ = \ 0,
$$
so the system splits into two subsystems $T_1$ and $T_2$, defined by
\be T_1 : = \left\{  \begin{array}{ccc}
t_1 & = & t_2
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \gamma \sqrt \om
\end{array}
\right. \nonumber
\ee
and
\be T_2 : = \left\{  \begin{array}{ccc}
t_1^2 + t_2^2 & = & 1
\\ & & \\
t_1^{-1} + t_2^{-1} & = & \gamma \sqrt \om
\end{array}
\right. \nonumber
\ee
The system $S_1$ gives the symmetric solution. The condition $\om >
\om_0$ emerges by the prescription $t_1, t_2 < 1$, so point {\em 1.} is
proven. 

Let us consider the system $T_2$. Multiplying the second equation by $t_1t_2$ 
one obtains
\be \label{summa}
t_1 + t_2 \ = \ \gamma \sqrt \om t_1 t_2,
\ee
thus squaring both sides, using the first equation of $T_2$ and
imposing positivity,   we have
$$
t_1 t_2 \ = \ \f{1 + \sqrt{1 + \gamma^2 \om}} {\gamma^2 \om}
$$
and so, by \eqref{summa}
\be \label{strauss}
t_1 + t_2 \ = \ \f{1 + \sqrt{1 + \gamma^2 \om}}{\gamma \sqrt \om}.
\ee
By \eqref{strauss} and the first equation of $S_2$, we finally get
\be \label{kahn}
t_{1,2} \ = \ \f{ \beta \pm \sqrt{2 - \beta^2}} 2
\ee
with $\beta : =  \f{1 + \sqrt{1 + \gamma^2 \om}}{\gamma \sqrt
  \om}$. The positivity of the quantity under square yields
the condition $\om > \om^*$. By \eqref{kahn} the proof of 2. is complete.
\end{proof}

The evaluation of the action $S_\om$ on the ground states too can be
explicitly computed. 
One obtains
\begin{prop}
Fix $\mu = 1$. Then,
\be \label{dcubic}
d (\om) \ = \ \left\{ \begin{array} {ccc}
\f {2 \lambda} 3 \om^{\f 3 2} \left( 2 - \f 6 {\gamma
      \sqrt \om} + \f 8 {\gamma^3 \om^{\f 3 2}}   \right), & & \om_0 <
  \om \leq \om^* \\ & & \\
 \f \lambda 3 \om^{\f 3 2} \left( 4 - \f{1 + \sqrt{1 + \gamma \sqrt \om}}
   {\gamma \sqrt \om} \left( 2 + \f {1 + \sqrt{1 + \gamma \sqrt \om}}
   {\gamma^2 \om} \right)\right), & & \om > \om^*.
\end{array}
\right. \ee
\end{prop}
\begin{proof}
We preliminarily recall that 
\be \label{intbase}
\int_0^\infty \f{dx}{\cosh^4( \sqrt \om (x-z))} \ = \ \f 1 {3 \om^{\f
    32 }} ( 2 + 3t - t^3), \qquad t : = \tanh (\sqrt \om z).
\ee
Since $d (\om) \ = \ \f \lambda 4 \| \psi_\om \|_4^4$, where
$\psi_\om$ is any ground state with frequency $\om$,
the computation for the case $\om \leq \om^*$ can be immediately
performed and provides the first formula in \eqref{dcubic}.
 
For $\om > \om^*$, using \eqref{intbase} gives
\be
d (\om) \ = \ \lambda \om^{\f 3 2} \left[ \f 4 3 - (t_1 + t_2) +
  \f{(t_1 + t_2)^3} 3   \right].
\ee
By \eqref{strauss} the computation can be completed up to obtaining
the second formula in \eqref{dcubic}, and this completes the proof.



\end{proof}


\end{document}





exact solvability of the model holds for the standard
nonlinear Sc




One of the most celebrated features of the point interactions (PI) lies in their
capability of supplying exactly solvable models (\cite{[AGHH],[AK]}). For this reason, 
PI have been widely employed to construct toy models and for pedagogical
purposes. Nonetheless, their exactness proves useful also when PI are involved in 
models of treal physical systems: the more a physically relevant quantity
 (e.g. energy spectrum or time evolution)
is explicitly known, the more information can be extracted. 

In particular, PI fit well the needs of modeling the so-called defects, 
namely, small inhomogeneities in a medium where a wave propagates, under the hypothesis
that the details of the internal structure of the inhomogeneity are not relevant,
so that its action can be modeled as concentrated at a point.
More precisely, the smallness of the inhomogeneity is to be evaluated with respect to the
typical wavelength of the incoming waves, or equivalently, in the case of a quantum
system, to the width of the wave function. In this paper we would like
to address the analysis of effects of the interaction between
nonlinearity and point defects in the behaviour of solutions of
nonlinear Schr\"odinger (NLS) equation. We prefer not to enter in a
description of the vast field of application of the NLS equation, from
the theory of integrable systems and inverse scattering to the
propagation of amplitude envelope of waves. We quote just two relevant
applications of the NLS as an effective model for real physical
systems: dynamics of Bose-Einstein condensates and laser beam
propagation in nonlinear (Kerr) media. In both cases it is physically
meaningful to consider the propagation of NLS waves in the presence of
defects. 
In particular, the recent spectacular development of both theoretical research
and experimental technology involving Bose-Einstein condensates
(BEC) (see \cite{stringari} and references therein, and \cite{[BKP],
  [BKP2],bcstv}) provides point interactions with a wide range of
  applications.  

As widely known, in current experiments the formation of a BEC is
induced in bounded region of spaces, usually delimited
by magnetic and/or optical traps. In such situations, the 
condensate lies in a one-particle quantum state, whose corresponding wavefunction is 
characterized as the minimizer of the Gross-Pitaevskii functional. When the trap is removed, 
the wavefunction of the BEC spreads out according to the evolution prescribed
by
the cubic Schr\"odinger equation
\be \label{gp}
i \partial_t \psi (t,x) \ = \ - \partial_x^2 \psi (t,x) + \alpha | \psi (t,x) |^2 \psi (t,x),
\ee
where we denoted by $\psi$ the wave function of the condensate, and the space variable
$x$ belongs to $\erre$, $\erre^2$ or $\erre^3$ according to the fact that we are modeling
a {\em cigar-shaped} or a {\em disc-shaped} or a genuinely three-dimensional BEC. 
We recall that the nonlinearity carries the information that, even though its state is
an actual one-particle state, the condensate consists of a large number of interacting
particles (in
the experimentally realized condensates, at least thousands); the fact that the nonlinearity
is cubic means that the dynamical effects of the two-particle interactions overwhelm the
effects of many-body collisions. The strength of the nonlinear term, given by the
constant $\alpha$, is proportional to the scattering length of the
two-body interaction
between the particles.
Here we do not summarize the progress in the derivation of \eqref{gp} as an
effective equation for a many-body quantum system. The interested reader is referred to
\cite{esy1,esy2,esy3} for the three-dimensional problem, \cite{kss} for
the two-dimensional case, and to \cite{agt,abgt,ab} for the case of 
cigar-shaped condensates. In the following we focus on this last case, in which, on one hand,
the nonlinearity is milder, while, on the other hand, the family of point interactions is
richer.

A natural question in this context is the following: what happens when a wave (i.e. a 
condensate) is sent against a defect? One would guess (and it has been shown for some
models, see e.g. \cite{gow,hmz,p}) that the incoming wave (think
of a solitary wave) splits
into a reflected wave, a transmitted wave and a captured component. Similar results have
been proven for propagation on graphs also (\cite{[ACFN1]}), in the case
of a repulsive vertex, where no capture occurs. The regime of fast
soliton has been also investigated in \cite{dh}.
Indeed,
it seems reasonable to conjecture that a capture can occur only if a nonlinear stationary
state exists. Since equation \eqref{gp} is dispersive,
the presence of a nonlinear stationary state (or more than one) must be related to the
defect. This is the reason why such possible stationary states are called {\em defect modes}.
Even though at this stage it is an unproven fact, it is plausible to
link the persistence of a captured wave with some sort of stability
(in a sense to be made precise) of the defect mode. For this reason
the interest in 
determining the stability of the defect modes lies not only in the problem itself, but
extends to models of reality too.

As a short review on results on stability and instability of defect modes 
in the presence of a power nonlinearity $| \psi |^{2 \mu} \psi$,
we recall  results proved in
\cite{reika,giannigianni,lacozza}, where the
effects of a $\delta$-like 
defect are analysed. The first cited work deals with an attractive defect,
and shows that, for any frequency $\om$ above the proper frequency of the
unique bound state of the delta potential, there is a unique defect mode that oscillates
at frequency $\om$. It turns out that the wavefunction of such a defect mode is nothing but the nonlinear
deformation of the linear bound state. The stability (more exactly, the 
{\em orbital} stability, see Definition \ref{os}) of such a mode depends on $\mu$ and $\om$:
if $\mu \leq 2$, then the defect mode is stable for any $\om$; if $\mu > 2$, then it
becomes unstable at high frequencies. References \cite{giannigianni,lacozza}
extend the analysis to a repulsive delta-like defect.
The situation becomes more involved in the case of a more singular defect, for instance,
the so-called $\delta^\prime$ defect. The following sections are
devoted to this case. For more details see also the comprehensive
review \cite{[ANS]} and the forthcoming paper \cite{[AN2]}. 

The established theoretical framework for the study of stability is provided by
Grillakis-Shatah-Strauss' theory (\cite{gss1,gss2}) or,
alternatively, by Lions' concentration-compactness method
(\cite{l1,l2} and \cite{c} for a review). The occurrence of
bifurcation in the ground state has been investigated in \cite{jw} and
more recently in \cite{supersacchetti,fukusacchetti,kirr,kevrekidis,tuoc}.

\section{Results}

\subsection{The $\delta^\prime$ "potential".}
The so-called $\delta^\prime$-defect, with strength $- \gamma$, located (just to be definite) at zero, 
is defined imposing the boundary 
condition
\be \label{bc}
\psi (0+) - \psi (0-) = - \gamma \psi^\prime (0+) = -\gamma \psi^\prime (0-)
\ee
to the solutions to \eqref{gp} (see \cite{abd,[EG]}). The parameter $\gamma$ is real; when positive,
the defect is called {\em attractive}, otherwise {\em repulsive}. More formally, one
defines a {\em linear Hamiltonian operator} $H_\gamma$ as the operator that acts as
$- \partial_x^2$ on functions in $H^2(\erre^+) \oplus H^2(\erre^+)$ satisfying 
\eqref{bc}. Note that the only continuous elements of the domain of
$H_{\gamma}$ have a vanishing derivative at the origin.  The operator
$H_\gamma$ is a self-adjoint operator with the following spectral
features: singular continuous spectrum is empty, absolutely
continuous 
spectrum is given by the positive halfline and point spectrum is empty in the
repulsive case, and coincides with $\{ - 4 / \gamma^2 \}$ in the attractive case.
In the last case the corresponding (non-normalized) eigenfunction is 
$$ \varphi_\gamma (x) \ = \ \epsilon (x) e^{-\f 2 \gamma |x|},$$
where we denoted the sign function by $\epsilon$. Notice that $\varphi_\gamma$ is odd. 
The quadratic form $F_\gamma$ associated to $H_\gamma$ is defined
on the domain $Q : = H^1 (\erre^+) \oplus H^1 (\erre^-)$ (we stress that $Q$ is independent
of $\gamma$) and reads
$$
F_\gamma (\psi) \ = \ \| \psi^\prime \|^2 - \gamma^{-1} | \psi (0+) - \psi (0-)
|^2,
$$
where we made the following abuse of notation
$$
\| \psi^\prime \|^2 : = \lim_{\ve \to 0+} \int_\ve^{+\infty} | \psi^\prime (x) |^2
dx + \lim_{\ve \to 0+} \int^{-\ve}_{-\infty} | \psi^\prime (x) |^2
dx,  
$$
that will be extensively repeated.

At variance with the delta potential, the Schr\"odinger operator with a $\delta^\prime$ interaction cannot be derived from a form sum, because the $\delta^\prime$ is not small with respect to the laplacian. 
Nevertheless it can be obtained as the norm-resolvent
limit of the sum of three $\delta$ potentials (\cite{cs}, \cite{enz}) with a fine tuned rescaling, defined as follows
$$[H_\gamma + \nu]^{-1} \ = \ \lim_{\ve \to 0}
\left[- \partial_x^2 - \left(\f 1 \gamma + \f 1 {2 \ve} \right) 
\delta (x - \ve) - \left( \f 1 \ve + \f \gamma {2 \ve^2} 
  \right) \delta (x) -  \left(\f 1 \gamma + \f 1 {2 \ve} \right) 
\delta (x - \ve) + \nu \right]^{-1}
$$
for any $- \nu$ in the resolvent set of $H_\gamma$.\par\noindent
Moreover,
since any delta potential, in its turn, can be approximated by a strong limit of rescaled regular
potentials, then it is possible to interpret a $\delta$-prime
potential as the suitable 
limit of rescaled, well-behaved potentials.  
\begin{figure}
\begin{center}
%{}{}\scalebox{0.40}{\includegraphics{approx.pdf}}
{}{}\scalebox{0.40}{\includegraphics{approx.eps}}
\caption{A regular approximation for an attractive $\delta$-prime
  potential centred at zero.
To obtain an approximation for a repulsive $\delta$-prime, one must
reverse the central well.}  
\end{center}
\end{figure}
%\end{minipage}
%\begin{minipage}{0.48\textwidth}
Let us remark that if $\psi$ belongs to the {\it operator} domain of
$H_\gamma$, then the form associated to $H_{\gamma}$ has the expression
$$
 (\psi, H_\gamma \psi) \ = \ \| \psi^\prime \|_2^2 
- \gamma | \psi^\prime (0+)|^2 ,
$$
which explains the questionable name of $\delta^\prime$.
%\end{minipage}
\subsection{Combining nonlinearity and defect} Once constructed the
operator  $H_\gamma$, the 
evolution in the presence of both a generic power nonlinearity 
and a defect is defined by
\be \label{gpdefect}
i \partial_t \psi (t,x) \ = \ H_\gamma \psi (t,x) + \alpha | \psi (t,x) |^{2 \mu} 
\psi (t,x).
\ee
For such equation it is possible to prove global well-posedness if
$\mu < 2$ (\cite{an1}), 
local  well-posedness if $\mu \geq 2$ (\cite{[AN2]}), and to provide examples of blow-up
for this last case (\cite{as}). However, until the solution exists, $L^2$-norm and
energy
$$ {\mathcal E}(\psi) = \f 1 2 \| \psi^\prime \|^2 - \f 1 {2 \gamma}
| \psi (0+) - \psi (0-) |^2 - \f \lambda {2 \mu + 2}  \| \psi \|_{2
  \mu + 2}^{2 \mu + 2}
$$
are conserved by time evolution.

Thanks to the existence of a conserved energy it is possible to introduce a notion of
{\em nonlinear ground state}: intuitively, one would define it as a minimizer of the energy among
the wavefunctions endowed with the same $L^2$-norm, as this is the definition that naturally
extends the more familiar notion of linear ground state. As in the linear case, it is meaningful to search for {\it stationary states} of \eqref{gpdefect}, i.e. solutions of the form
\be 
\psi(x,t) = e^{i\omega t} \psi_{\omega}(x)\ .
\ee
The amplitudes $\psi_{\omega}$ are solutions of the stationary equation

\be \label{stateq}
H_\gamma \psi_{\omega} + \om \psi_{\omega} - \lambda | \psi_{\omega} |^{2 \mu} \psi_{\omega} \ = \ 0.
\ee

This leads to the introduction of the so-called {\em action functional}
\be \label{som} 
S_\om ( \psi ) \ = \ {\mc E} (\psi) + \f \om 2 \| \psi \|^2,
\ee
defined on the energy domain $Q$. 
It is immediate indeed that Euler-Lagrange equations for $S_{\om}$ are given just by \eqref{stateq}. 
Note that the action (and the energy as well) is not bounded from below on $Q$.
To overcome this problem, a {\em ground state} $\psi_{\om}$ is usually defined as a minimizer of $S_\om$ constrained
on the {\em Nehari manifold} 
$$
I_\om (\psi) =
S_\om^\prime ( \psi ) \psi = ( \psi, H_\gamma \psi - \lambda
| \psi |^{2 \mu}  \psi + \om \psi) \ = \
0.
$$
The above set is a codimension one manifold that obviously contains all stationary points of 
$S_\om$, and it tuns out that on it the action is bounded from below.\par\noindent 
It is a byproduct of the Grillakis-Shatah-Strauss theory on stability
of stationary states (\cite{gss1}, \cite{gss2}) 
applied to minimizers of $S_\om$ the relation between the constrained variational problem for ${\mathcal E}$ and $S_\om$: a minimizer $\psi_\omega $ of the action on the
  Nehari manifold is a minimizer of the energy among the function with
  the same $L^2$-norm $\|\psi_\omega\|^2$ {\em if and only if it is stable} (in the sense
  of Definition \eqref{os}).

The following preliminary result is obtained through variational techniques (we remove the subscript $\omega$ from $\psi_{\om}$ when not needed to avoid ambiguity):
\begin{theorem} \label{prel}
For any $\om  > \f 4 {\gamma^2}$ there exists at least one minimizer of $S_\om$ among
all functions on the Nehari manifold. Furthermore, the minimizer  solves
the stationary Schr\"odinger equation with defect:
\be \label{stazz}
H_\gamma \psi + \om \psi - \lambda | \psi |^{2 \mu} \psi \ = \ 0.
\ee
\end{theorem}
The line of the proof is standard, except that: first, the functional
space of reference $Q$ is larger than $H^1 (\erre)$; second, the problem is one-dimensional,
so that one must cope with a lack of compactness when passing from weak convergence in
$Q$ to strong convergence in $L^p$; third, the boundary condition to be reconstructed is non
standard. A complete proof is in \cite{[AN2]}.

An important point about Theorem \ref{prel} is that, in order to find the ground states,
it suffices to determine which one among the solutions of
\eqref{stazz} has 
least action. This can be made directly, as
the solutions to equation \eqref{stazz} can be explicitly found. Moreover the minimum is constrained to a finite codimension (one in this case) manifold, an information which is important for stability issues.

\subsection{Symmetry breaking}
Equation \eqref{stazz} can be rephrased as follows:
\be \label{reph}
-\partial_x^2 \psi + \om \psi - \lambda  | \psi |^{2 \mu} \psi \ = \ 0,
\ee
with $\psi \in H^2 (\erre^+) \oplus H^2(\erre^-)$ satisfying the
boundary condition \eqref{bc}. The only solutions to \eqref{reph} that
vanish at infinity are constructed
 by gluing together two pieces of a solitary
wave for the NLS, namely
\begin{equation*}
\psi^{x_1, x_2}_{\om, \pm} (x) = \left\{ \begin{array}{cc}
\pm \lambda^{-\f 1 {2\mu}} (\mu + 1)^{\f 1
  {2\mu}} \om^{\f 1 {2\mu}} \cosh^{-\f 1 \mu} [\mu \sqrt \om ( x - x_1)], & x<0 \\
\lambda^{-\f 1 {2\mu}} (\mu + 1)^{\f 1
  {2\mu}} \om^{\f 1 {2 \mu}} \cosh^{-\f 1 \mu} [\mu \sqrt \om ( x - x_2)], & x>0
\end{array}
\right.
\end{equation*}    
where the parameters $x_1$ and $x_2$ are to be adjusted so that
\eqref{bc} is satisfied. Now, it is immediately seen by \eqref{som} that due to
contribution of the point interaction energy, one has
$$S_{\om} (\psi^{x_1, x_2}_{\om, -}) \ < \ S_\om (\psi^{x_1, x_2}_{\om, +})$$
so we can restrict the search for minimizers to the functions $\psi^{x_1,
  x_2}_{\om, -}$. 

For any such function, the boundary condition \eqref{bc} 
translates into the system
\begin{equation} \label{tsystem}
 \left\{
\begin{array}{ccc}
t_1^{2 \mu} - t_1^{2 \mu + 2} & = &  t_2^{2 \mu} - t_2^{2 \mu + 2} \\ 
t_1^{-1} + t_2^{-1} & = & \gamma \sqrt \om
\end{array}
\right. ,  \ 0 \leq t_i = | \tanh (\mu \sqrt \om x_i)| \leq 1,
\end{equation}
whose solutions can be depicted as the intersection of the full and
the dashed lines in Figure 2.
\begin{figure} \label{fig1}
\begin{center}
%{}{}\scalebox{0.50}{\includegraphics{bif2-bis.pdf}}
{}{}\scalebox{0.50}{\includegraphics{bif2-bis.eps}}
\caption{The full lines represent the solutions to the first equation
  in
\eqref{tsystem}: they consist of the line $0 \leq t_1 = t_2 \leq 1$, and of a
curve, that is concave if $\mu$ is not too small. The dashed
lines represent the solutions to the second equation of
\eqref{tsystem}: they consist of a family of hyperbola parametrized
by $\om$.
}
\end{center}
\end{figure}
One immediately finds that for $\f 4 {\gamma^2} < \om \leq \f 4
{\gamma^2} \f {\mu + 1} \mu$ the unique solution is given by $t_1 =
t_2 = \f 2 {\gamma \sqrt \om}$, that corresponds to an antisymmetric
 stationary state 
$\psi_\om^{y, -y}$, where
$$
y = x_1 = - x_2 = \f 1 {2 \mu \sqrt \om} \log \f {\gamma \sqrt \om + 2}
{\gamma \sqrt \om  - 2}. $$ 
At $ \om = \f 4
{\gamma^2} \f {\mu + 1} \mu$ two new solutions arise, giving birth to two
new branches of stationary states that persist for $\om >\f 4
{\gamma^2} \f {\mu + 1} \mu$; they correspond to the
couple of asymmetric stationary states $\psi_\om^{y_1, - y_2}, \, 
\psi_\om^{y_2, - y_1}$, with both $y_1$
and $y_2$ positive but, except in the cubic case $\mu = 1$, not in explicit form. 
A direct computation yields, for these values of $\omega$,
$$ S_\om (\psi_\om^{y_1, -y_2}) \ < \ S_\om (\psi_\om^{y, - y})\ .$$
We conclude that with the growth of the frequency $\omega$ there exist {\em two} branches of asymmetric ground states which bifurcate from the branch of (anti)symmetric ones. We are then in the presence of a spontaneous symmetry breaking of the set of ground states.

\subsection{Stability: a pitchfork bifurcation}
The study of the stability for such a system can be made by applying
the Grillakis-Shatah-Strauss theory (\cite{gss1}, \cite{gss2}). 
This theory provides sufficient conditions for the {\em{orbital
    stability}} of stationary states, which is stability ``up to the symmetries". Roughly speaking, the notion of
orbital stability coincides with the ordinary Ljapunov stability {\em
  for orbits instead of states}, where orbits are to be understood 
with respect to a symmetry group. In our case the symmetry group is $U(1)$, corresponding to the well known phase invariance of the NLS, which persists in the presence of point perturbation too. 
So, a stationary state $\psi_\om$ is
said to be {\em orbitally stable} if at any time a solution to
\eqref{gpdefect} remains arbitrarily close {\em to the orbit} $\{
e^{i \theta} \psi_\om, \, \theta \in [0, 2 \pi)  \}$, provided that it
started sufficiently close to it. More rigorously,
\begin{definition} \label{os}
A stationary state $\psi_\om$ is called {\em orbitally stable} if
for any $\ve > 0$ there exists a $\sigma > 0$ s.t.
$$ \inf_{\theta \in [0, 2 \pi)} \| \psi_0 - e^{i \theta} \psi_\om \|_Q
\leq \sigma \Rightarrow \sup_{t>0} \inf_{\theta \in [0, 2 \pi)} 
\| \psi_t - e^{i \theta} \psi_\om \|_Q
\leq \ve,
$$
where $\psi_t$ is the solution corresponding to the initial condition
$\psi_0$.

A stationary state is called {\em orbitally unstable} if it is not
orbitally
stable.

\end{definition}

The Grillakis-Shatah-Strauss theory (see \cite{gss1,gss2})
 carries out a deep investigation of the orbital stability of stationary states of (infinite dimensional) hamiltonian systems with symmetries, generalizing previous work by the same authors and independently by Michael Weinstein 
(\cite{[W1],[W2],[W3]}).\par\noindent
They succeed in giving sufficient conditions for stability and instability by studying  second-order approximation of the action (linearization) around a stationary
state, and carefully controlling the nonlinear remainders exploiting symmetries and conservation laws. In the present situation, as it is well known, one gets a hamiltonian system from NLS equation passing to real variables $(\eta, \rho)=({\rm Re} \psi, {\rm Im} \psi)$. We confine ourself to a brief operative summary of the method, and so we omit the (however important) connection with hamiltonian systems referring to the original literature for details.\par\noindent
Neglecting higher order terms, one has for the action expanded around the stationary state $\psi_{\omega}$ (we omit other superscripts for simplicity)
$$
S_\om (\psi_\om + \eta + i \rho) \cong S_\om (\psi_\om) + \f 1 2 
\left( S^{\prime \prime}(\psi_\om) \left( \begin{array}{c} \eta \\ \rho
\end{array} \right), \left( \begin{array}{c} \eta \\ \rho
\end{array} \right) \right).
$$
The Hessian operator $S_\om^{\prime \prime}(\psi_{\om})$ can be
represented in matrix form as (we implicitly introduced the representation of a function 
$\eta + i \rho$ as the real vector function $(\eta, \rho)$)

$$S^{\prime \prime}_\om (\psi_\om)
\ : = \  \left( \begin{array}{cc}
 L_1 & 0
\\
0 & L_2
%\left( \begin{array}{c} \eta_t \\ \rho_t
\end{array} \right),
$$
where $L_1$ and $L_2$ are two selfadjoint operators with $D(L_1)=D(L_2)=D(H_{\gamma})$ given by
\begin{eqnarray*}
L_1 & = &   H_\gamma +
\om -  \lambda (2 \mu + 1) | \psi_\om |^{2 \mu} \\
L_2 & = &   H_\gamma +
\om -  \lambda | \psi_\om |^{2 \mu}
.
\end{eqnarray*}
Now, were $S^{\prime \prime}_\om (\psi_\om)$ a positive operator, the (linear)
stability of $\psi_\om$ would be immediately established, as
the situation would be analogous to what happens for a
classical particle in a potential well. However this is cannot be the
case. First of all, the operator $S_\om^{\prime \prime} (\psi_\om)$ is
endowed with a non trivial kernel that consists of the linear span of
$(0, \psi_\om)$, due to the symmetry. Second, recall that every ground state 
$\psi_\om$ is a minimizer only on the constraint provided by the
Nehari manifold, which has codimension one. On the space orthogonal to
the Nehari manifold,
$S_\om^{\prime \prime} (\psi_\om)$ is surely negative, as 
$$ (\psi_\om, S_\om^{\prime \prime} \psi_\om) \ < \ 0. $$
It follows that there exists a cone on which 
$S_\om^{\prime \prime} (\psi_\om)$ is actually negative. 

\noindent
Nevertheless, it is possible that the dynamical
constraints given by the conservation laws prevent the wave function
from further evolving far inside that cone, finally forcing the solution
to remain close to the orbit of the ground state. The Grillakis-Shatah-Strauss theory 
establishes that this is the case if a certain number of conditions are satisfied. 
In its easiest version, such a set of
conditions can be collected as follows

\noindent
$i)$ {\em Spectral conditions:} 
\begin{enumerate}
\item
${\rm{Ker}} L_2 \ = \ {\rm{Span}} \{ \psi_\om \}$,
\item  $L_2 \geq 0$,
\item ${\rm{Ker}} L_1 \ = \ \{ 0 \}$,
\item $L_1$ has exactly {\em one} negative eigenvalue.
\end{enumerate}

\noindent
$ii)$ {\em Vakhitov-Kolokolov's criterion} (\cite{vk}): $$\f {d \| \psi_\om \|_2}{d\om} 
> 0,$$
that, since $\f {dS_\om (\psi_\om)} {d \om} = \f 1 2 \| \psi_\om
\|_2^2$, is equivalent to
\be \label{vk}
 \f {d^2S_\om (\psi_\om)} {d \om^2} > 0.
\ee

In the case of interest, conditions $i)$ and $ii)$ are verified except for the
the stationary states in the branch $\psi_\om^{y, -y}$ with $\omega>\f 4 {\gamma^2} \f {\mu + 1} \mu $, where more a more sophisticated version of conditions $i)$ and $ii)$ is needed, again provided by the Grillakis-Shatah-Strauss theory 
(see \cite{gss2}).   



The
results on stability can be summed up as follows.
\begin{theorem} For any $\mu > 0$
\begin{enumerate}
\item If $\om < \f 4 {\gamma^2} \f {\mu + 1} \mu$, then the 
unique (up to a phase) ground state $\psi_\om^{y, -y}$ is orbitally
stable.
\item If $\om > \f 4 {\gamma^2} \f {\mu + 1} \mu$, then the stationary
  state $\psi_\om^{y, -y}$ is orbitally
unstable.
\end{enumerate}
For $0 \leq \mu \leq 2$, $\om > \f 4 {\gamma^2} \f {\mu + 1} \mu$, the
two ground states $\psi_\om^{y_1, -y_2}$, $\psi_\om^{y_2, -y_1}$ are
orbitally stable.

For $\mu > 2$ there exist $\om_1 > \f 4 {\gamma^2} \f {\mu + 1} \mu$,
$\om_2 > \om_1$, such that, if $\f 4 {\gamma^2} \f {\mu + 1} \mu < \om
< \om_1$ then  $\psi_\om^{y_1, -y_2}$ and $\psi_\om^{y_2, -y_1}$ are
orbitally stable; if $\om > \om_2$, then  $\psi_\om^{y_1, -y_2}$ and
 $\psi_\om^{y_2, -y_1}$ are
orbitally unstable.
\end{theorem}
The bifurcation diagrams for the system are portrayed in Figures
3 and 4.
\begin{figure} \label{figA}
\begin{center}
%{}{}\scalebox{0.50}{\includegraphics{bifstab-bis.pdf}}
{}{}\scalebox{0.50}{\includegraphics{bifstab-bis.eps}}
\caption{Bifurcation diagram for $\mu \leq 2$.
The full line denotes stable stationary states, while the dashed line
represents unstable stationary states. Notice that ground states are always stable.
}
\end{center}
\end{figure}
\begin{figure} \label{figB}
\begin{center}
%{}{}\scalebox{0.50}{\includegraphics{bifuncertainright-bis.pdf}}
{}{}\scalebox{0.50}{\includegraphics{bifuncertainright-bis.eps}}
\caption{Bifurcation diagram for $\mu > 2$.
We have no results for the interval $(\om_1, \om_2)$, but we conjecture that
it is always possible to choose $\om_1 = \om_2$.
}
\end{center}
\end{figure}

\section{Proof of stability}
The content of this section is technical. Here we
we give a proof of the stability of all ground states in the case $\mu \leq 2$. Under
such a restriction, every ground state 
satisfies the Vakhitov-Kolokolov's criterion. The proof we present differs from the
one given in \cite{[AN2]}, as it does not use the Grillakis-Shatah-Strauss theory and so it does not refer to linearization. We 
decided to include in this report such a technical part in order to
convey some information on the method of proofs and on the techniques employed.
An analogous analysis is given for the case of a NLS with $\delta$ interaction in \cite{reika}, and both are inspired by \cite{fibich}.\par\noindent
In order to proceed we need some preliminary definitions and results.

First, the definition of orbital stability can be reformulated using the
notion of {\em orbital}  {\em neighbourhood}.
\begin{definition}
The set
$$ U_\eta (\phi) : = \{ \psi \in Q, \, {\rm s.t.} \, \inf_{\theta \in [0, 2 \pi)}
\| \psi - e^{i \theta} \phi \|_Q \leq \eta \}$$
is called the {\em orbital neighbourhood with radius $\eta$ of the function $\phi$}.
\end{definition}

It is convenient to introduce a function that associates to any
frequency $\om > \f 4 {\gamma^2}$ the value of the minimum attained by
$S_\om$ evaluated on functions in the Nehari manifold. Namely,
\begin{eqnarray*}
d & : & \left( \f 4 {\gamma^2}, + \infty \right) \to \erre \\
 & & \om \mapsto \min \{ S_\om (\phi), \phi \in Q, I_\om (\phi) = 0 \}.
\end{eqnarray*}

It is then important to stress other points that we did not mention
explicitly so far.
\begin{remark} \label{tredue} \

\noindent
\begin{enumerate}
\item In the energy space $Q$ the following norm is defined:
\begin{equation*} \begin{split}
\| \phi \|_Q^2 \ : = & \ \| \phi \|_2^2 + \lim_{\ve \to 0+}
\int_\ve^{+\infty} | \phi^\prime|^2 \, dx + 
 \lim_{\ve \to 0+}
\int^{-\ve}_{-\infty} | \phi^\prime|^2 \, dx
\end{split}
\end{equation*}
\item For any $\theta \in [0, 2 \pi)$, any $\phi \in Q$, one has
$S_\om (e^{i \theta} \phi) =
  S_\om (\phi)$. As a consequence, if $\psi_\om$ is a ground state,
  then all the functions $e^{i \theta} \psi_\om$ in its orbit are
  ground states too. In the proof of Theorem \eqref{teo:stability} we
  will make the phase explicit by denoting 
$$\psi_{\om}^{x_1, x_2, \theta} \ : = \ e^{i \theta} \psi_{\om}^{x_1, x_2, 0}.$$

\end{enumerate}

\end{remark}



The result we need to go through the proof are summarized in the
following Proposition. Their proof is contained in \cite{[AN2]}.

\begin{prop} \label{tretre} \

\noindent
\begin{enumerate}
\item For any function $\phi$ in the Nehari manifold one has $S_\om (\phi) =
  \widetilde S (\phi)$, where $\widetilde S$ is the functional defined
  by
$$ \widetilde S (\phi) : = \f {\lambda \mu} {2 \mu + 2} \| \phi \|_{2
  \mu + 2}^{2 \mu + 2}.$$
\item Any minimizer $\psi_\om$ of the functional $S_\om$ on the Nehari
  manifold minimizes also the functional $\widetilde S$ on the region
  $I_\om \leq 0$.
\item Following Fibich and Wang (\cite{fibich}) we recall that 
the map
\be \begin{split}
\nonumber
\widehat \om : Q \ \longrightarrow & \ \erre, \qquad
\phi \ \mapsto  \ d^{-1} \left( \f \lambda 2 \f {\mu} {\mu + 1} 
\| \phi \|_{2 \mu + 2}^{2 \mu + 2}
\right)
\end{split}
\ee
is well-defined. Notice that $\widehat \om$ maps a function $\phi$ into the
frequency of a ground state having the same $L^{2 \mu + 2}$-norm as
$\phi$. Such a ground state may not be unique, but the $L^{2 \mu +
  2}$-norm always is.



\item If $\psi_\om$ minimizes $S_\om$ on the Nehari manifold $I_\om = 0$, then $\psi_\om$
minimizes 
$S_\om$ on the set 
 $
\{ \phi \in Q, \, \| \phi \|_{2 \mu + 2 } = \| \psi_\om \|_{2 \mu + 2}
\}.
$

\item For any $\om > 0$, the function $\chi_{[0, + \infty)} \psi_{\om}^{0,0}$ minimizes 
the functional
$$
S_\om^0 (\phi) : = \f 1 2 \| \phi^\prime \|_2^2 + \f \om 2  \| \phi
\|_2^2 - \f \lambda {2 \mu + 2} \| \phi \|_{2 \mu + 2}^{2 \mu + 2} 
$$
among the functions in $Q$ that satisfy
$$
I_\om^0 (\phi) : =  \| \phi^\prime \|_2^2 + \om   \| \phi
\|_2^2 - \lambda  \| \phi \|_{2 \mu + 2}^{2 \mu + 2} \ = \ 0. 
$$

\item If $\mu \leq 2$, then any ground state satisfies the
  Vakhitov-Kolokolov's condition \eqref{vk}.


%\item
%Given $c>0$,
%the set of the minima for the functional 
%$S_\om^0$ on the set
% $$
%\{ \psi \in Q, \, \| \psi \|_{2 \mu + 2 } = c
%$$
%reads
%\be \label{minima0}
%\{ e^{i \theta} \chi_+ \phi_0 , \ e^{i
%  \theta} \chi_- \phi_0, \, \theta \in [0, 2 \pi) \}.
%\ee
%\item Questo non lo capito
%$$d^{\prime \prime} \ = \ $$
\end{enumerate}
\end{prop}

Now we can prove the

\begin{theorem} \label{teo:stability}
If $1 \leq \mu \leq 2$, then any ground state is stable.
\end{theorem}

\begin{proof}
We specialize to the case with 
$\om > \f 4 {\gamma^2} \f {\mu + 1} \mu$, namely, beyond the 
frequency of bifurcation. In fact, for $\om < \f 4 {\gamma^2} \f {\mu + 1} \mu$,
this proof can be easily adapted and one recovers essentially
the argument given in \cite{reika}. 


Fix $\om_0 > \f 4 {\gamma^2} \f {\mu + 1} \mu$ and
suppose that the stationary solution $e^{i \om_0 t}
\psi_\om^{y_1, -y_2,0}$ is orbitally unstable. 
This means that there exists $\ve_0 > 0$ and a sequence
$\varphi_k \in U_{\f 1 k} (\psi_\om^{y_1, -y_2,0})$ such that
$$ \sup_{t \geq 0} \inf_{\theta \in [0, 2 \pi)} \| \varphi_k (t) -
 \psi^{y_1, -y_2,\theta}_{\om_0} \|_Q \geq \ve_0,$$
where $\varphi_k (t)$ is the solution to equation \eqref{gpdefect} with
initial data $\varphi_k$. 


With no loss of generality, we assume
\be \label{choiceps}
\ve_0 \ \leq \ 
\inf_{\theta \in [0, 2 \pi)}
 \| \psi^{y_1, -y_2,0}_{\om_0} - \psi^{y_2, -y_1, \theta}_{\om_0} \|_Q \ = \ 
\| \psi^{y_1, -y_2,0}_{\om_0} - \psi^{y_2, -y_1,0}_{\om_0} \|_Q .
\ee
Let $t_k$ be the smallest positive
time for which  
\be \label{smallest}
 \inf_{\theta \in [0, 2 \pi)} \| \varphi_k (t_k) -
\psi^{y_1, -y_2,\theta}_{\om_0} \|_Q = \f {\ve_0} 2,
\ee
and let us use the notation $\xi_k \ = \ \varphi_k (t_k)$.
By conservation laws,
%\be \label{prefibich}
%S_\om (\xi_k) \, \longrightarrow \, S_\om (\phi_0) 
%\ee
%for any $\om$, in particular
\be \label{particulare} \begin{split}
S_{\om_0} (\xi_k) \, = \,  & {\mathcal E} (\xi_k) + \f {\om_0} 2 \|
\xi_k \|_2^2 \, = \,
{\mathcal E} (\varphi_k) + \f {\om_0} 2 \| \varphi_k \|_2^2 \\
\,   \longrightarrow \, & {\mathcal E} (\psi^{y_1, -y_2,0}_{\om_0}) + \f {\om_0} 2 
\| \psi^{y_1, -y_2,0}_{\om_0} \|_2^2 \, = \,
S_{\om_0} (\psi^{y_1, -y_2,0}_{\om_0}) \, = \, d (\om_0). 
\end{split}
\ee
where we used the fact that, by construction, the sequence $\varphi_k$
converges to $\psi^{y_1, -y_2,0}_{\om_0}$ strongly in $Q$, that implies the
convergence of the energy and of the $L^2$-norm.

Let us denote $\om_k = \widehat \om (\xi_k)$. 
We recall the 
following result from \cite{fibich}, used in \cite{reika} also:
\be \label{fibich}
S_{\om_k} (\xi_k) - S_{\om_k} (\phi_0) \ \geq \ \f 1 4 d^{\prime \prime} (\om_0)
(\om_k - \om_0)^2
\ee 
where we denoted $\om_k \ = \ \om (\xi_k)$.
The fact that
the Vakhitov-Kolokolov's condition is satisfied (see (6)), together with
\eqref{fibich} and \eqref{particulare}, implies
 $\om_k \rightarrow \om_0$, and therefore, 
by the definition of the function $\widehat \om$, we have
\be \label{xil4}
\| \xi_k \|_{2 \mu + 2} \ = \ \left[ \f {2 \mu + 2} {\lambda \mu} 
 S_{\om_k} (\psi_{\om_k} ) \right]^\f 1 {2 \mu + 2}
\ \longrightarrow \ \left[ \f {2 \mu + 2} {\lambda  \mu} 
 S_{\om_0} (\psi_{\om_0} )   \right]^\f 1 {2 \mu + 2}  \ = \ 
 \| \psi^{y_1, -y_2,0}_{\om_0}\|_{2 \mu + 2}.
\ee

\n 
We define the sequence $\zeta_k : = \f {\| \psi^{y_2, y_1,0}_{\om_0} \|_{\mu + 2}}  
{\| \xi_k \|_{\mu + 2}} \xi_k$. By \eqref{xil4},
\be \label{convzQ}
\| \zeta_k - \xi_k \|_Q \ = \ \left| \f {\| \psi^{y_2, y_1,0}_{\om_0} \|_{\mu + 2}}  
{\| \xi_k \|_{\mu + 2}} - 1 \right| \|  \xi_k \|_Q \ \longrightarrow \
0.
\ee
As a consequence, $S_{\om_0} (\zeta_k) - S_{\om_0} (\xi_k)
\rightarrow 0$, so $S_{\om_0} (\zeta_k) \rightarrow S_{\om_0} (\psi_{\om_0})$. For
this reason, and as 
$\| \zeta_k \|_{2 \mu + 2} \ =
\ \| \psi^{y_1, -y_2,0}_{\om_0} \|_{2 \mu + 2}$,  
point (4) in Proposition \ref{tretre} implies that 
$\{ \zeta_k \}$ is a minimizing 
sequence for the problem
\be \nonumber
\min \{ S_{\om_0} (\psi), \ \psi \in Q \backslash \{ 0 \}, \
\| \psi \|_{2 \mu + 2} = \| \psi^{y_2, -y_1,0}_{\om_0} \|_{2 \mu + 2}
\}.
\ee
By Banach-Alaoglu theorem there exists a subsequence, whose elements we
denote by $\zeta_k$ too, that converges weakly in $Q$ and therefore
in $L^{2 \mu + 2}$. Let us call $\zeta_\infty$ its weak limit. 

First, notice that $\zeta_\infty \neq 0$. Indeed, were it zero, 
then weak convergence in $Q$ would imply
$\zeta_\infty (0 \pm) \rightarrow 0$, and therefore $S_{\om_0} (\zeta_k) -
S_{\om_0}^0 (\zeta_k) \rightarrow 0$, so 
$$
\lim_{k \to \infty} S_{\om_0}^0 (\zeta_k) \ = \ \lim_{k \to \infty}
S_{\om_0} (\zeta_k) \ = \ \f 
\lambda 2 \f \mu {\mu + 1} \| \psi^{y_1, -y_2,0}_{\om_0} 
\|_{2 \mu + 2}^{2 \mu + 2}.
$$ 
Then, employing the fact
that $\phi_0 : = 
\chi_{[0, + \infty)} \psi_{\om_0}^{0,0} (0) \neq 0$, points (4) and (5) in Proposition
\ref{tretre} yield
\be \nonumber
 \f
\lambda 2 \f \mu {\mu + 1} \| \psi^{y_1, -y_2,0}_{\om_0} 
\|_{2 \mu + 2}^{2 \mu + 2} \ = \
\lim S_{\om_0}^0 (\zeta_k) \ \geq \ S_{\om_0}^0 (\chi_+ \phi_0) \ > \ 
 S_{\om_0} (\chi_+ \phi_0) \  \geq \ \f \lambda 2 \f \mu {\mu + 1} 
\| \psi^{y_1, -y_2, 0}_{\om_0}
 \|_{2 \mu + 2}^{2 \mu + 2}, 
\ee
which is absurd. So it must be $\zeta_\infty \neq 0$.

We claim
that $I_{\om_0} (\zeta_\infty) \ = \ 0$. We proceed by contradiction.

\n
Suppose indeed that $I_\om
(\zeta_\infty) < 0$. 
Then, by point (2) in Remark \ref{tredue} and points (1) and (2) in
Proposition \ref{tretre}
we know that
the minimizers of  the functional $\widetilde S$ on
the region $I_{\om_0} \leq 0$ are given by
$\psi_{\om_0}^{y_1, -y_2, \theta}$ and $\psi_{\om_0}^{y_2, -y_1,
  \theta}$, for all $\theta \in [0, 2 \pi)$. 
Furthermore,  all such functions lie on the set $I_{\om_0}
= 0$.
As a consequence, recalling the definition 
of the
functional
$\widetilde S$, one obtains
$$\| \zeta_\infty \|_{2 \mu + 2} \ = \ \left[ \f{2 (\mu + 1)} {\lambda \mu}
\widetilde S (\zeta_\infty) \right]^{\f 1 {2 \mu + 2}}
\ > \  \left[ \f{2 (\mu + 1)} {\lambda \mu}
\widetilde S (\psi_{\om_0}^{y_1, -y_2, 0}) \right]^{\f 1 {2 \mu + 2}}
\ = \ \| \psi_{\om_0}^{y_1, -y_2, 0} \|_{2 \mu + 2}.$$
But this is not possible, as $\zeta_\infty$ is the
weak limit of functions having the same $L^{2 \mu +2}$-norm 
as $\psi^{y_1, -y_2,0}_{\om_0}$.

\n
On the other hand, suppose that $I_{\om_0} (\zeta_\infty) \ > \ 0$. 
By \eqref{particulare} and  \eqref{xil4}
\begin{equation*} \begin{split} &
\lim_{k \to \infty} I_{\om_0} (\xi_k) \ = \ 2  \lim_{k \to \infty}
S_{\om_0} (\xi_k) - \f {\lambda \mu} {\mu + 1}  \lim_{k \to \infty}
\| \xi_k \|_{2 \mu +
  2}^{2 \mu + 2} \\ \ = \ &  \ 2  
S_{\om_0} (\psi^{y_1, -y_2,0}_{\om_0}) - \f {\lambda \mu} {\mu + 1}  
\| \psi^{y_1, -y_2,0}_{\om_0}\|_{2 \mu +
  2}^{2 \mu + 2} \ = \  I_{\om_0} (\psi^{y_1, -y_2,0}_{\om_0}) \ = \
0. 
\end{split} \end{equation*}
Therefore, by \eqref{convzQ}, 
$$  \lim_{k \to \infty} I_{\om_0} (\zeta_k) \ = \ 0. $$
From the following inequality (see \cite{bl})  
\begin{eqnarray}
\| u_n \|_{p}^{p} - \| u_n - u_\infty \|_{p}^{p} 
- \| u_\infty \|_{p}^{p} &
\longrightarrow & 0, \quad \forall \, 1 < p < \infty. \label{breli1}
\end{eqnarray}
one easily has 
$$I_{\om_0} (\zeta_k
- \zeta_\infty) \longrightarrow - I_{\om_0} (\zeta_\infty) \ < \ 0.$$ 
As a consequence, eventually in $k$ we obtain $I_{\om_0} (\zeta_k
- \zeta) < 0$ and then, using point (2) in Proposition \ref{tretre}
\be \label{precontra}
\| 
\zeta_k - \zeta_\infty \|_{2 \mu + 2} \ > \ \| \psi^{y_1, -y_2,0}_{\om_0} 
\|_{2 \mu + 2}.
\ee
But
from \eqref{breli1}, and knowing that $\zeta_\infty \neq 0$, we have 
that the following inequality holds eventually in $k$
$$\|
\zeta_k - \zeta_\infty \|_{2 \mu + 2} \ \leq  \ 
\| \psi^{y_1, -y_2,0}_{\om_0} \|_{2 \mu + 2},$$
that contradicts \eqref{precontra}.

We conclude that $I_{\om_0} (\zeta_\infty)$ cannot be strictly
positive and, as we already proved that it cannot be negative,
it must vanish.

As a consequence, from point (2) in Proposition \ref{tretre} again, 
we get $\| \zeta_\infty \|_{2 \mu + 2} \geq
\| \psi^{y_1, -y_2,0}_{\om_0} \|_{2 \mu + 2}$. But, since
$\zeta_\infty$ is a weak limit, it must be
\be \nonumber 
\| \zeta_\infty \|_{2 \mu + 2} \ = \ \| \psi^{y_1, -y_2,0}_{\om_0} \|_{2 \mu + 2}.
\ee
This fact has the following relevant consequences:
\begin{itemize}
\item Owing to \eqref{breli1}, the sequence $\{ \zeta_n \}$ 
converges strongly to $\zeta_\infty$ in the topology of
  $L^{2 \mu + 2}$.

\item The sequence $\{ \zeta_k \}$ converges to $\zeta_\infty$ in the strong
topology of
  $Q$. Indeed, by the convergence of $S_{\om_0} (\zeta_k)$ to
  $S_{\om_0} (\zeta_\infty)$, the weak convergence in $Q$, and the strong
  convergence of $\{ \zeta_k \}$ in $L^{2 \mu + 2}$, we have
\be \label{qomega}
\| \zeta_k^\prime \|^2 + {\om_0} \| \zeta_k \|^2 \ \longrightarrow \ 
\| \zeta_\infty^\prime \|^2 + {\om_0} \| \zeta_\infty \|^2.
\ee
So the convergence is strong in the space $Q$ endowed with the norm
given by \eqref{qomega}, that is equivalent to the usual
$Q$-norm.

\item The sequence $\{ \xi_k \}$ also converges to $\zeta_\infty$ in the strong
topology of $Q$. Indeed,
applying \eqref{convzQ}, we have 
\be \label{poscontra}
 \| \xi_k - \zeta_\infty \|_Q \ \leq \ 
\| \xi_k - \zeta_k \|_Q + \| \zeta_k - \zeta_\infty
  \|_Q \ \longrightarrow  
\ 0.
\ee

\item The function $\zeta_\infty$ minimizes $S_{\om_0}$ with the
  constraint $I_{\om_0} \ = \ 0$, so, 
 either $\zeta_\infty \ = \ 
    \psi^{y_1, -y_2,\theta}_{\om_0}$ or $\zeta_\infty \ = \ 
     \psi^{y_2, -y_1, \theta}_{\om_0}$ for some value of $\theta$ in $[0, 2 \pi)$.


\end{itemize}
Let us suppose that $\zeta_\infty \ = \ 
    \psi^{y_1, -y_2,\theta}_{\om_0}$, for a certain value of $\theta$.
By \eqref{poscontra} we obtain $\xi_k \to  \psi^{y_1,
  -y_2,\theta}_{\om_0}$ strongly in $Q$, that
contradicts inequality \eqref{smallest}, and thus the assumption 
of the orbital instability
of the stationary state $\psi^{y_1, -
  y_2, 0}_{\om_0} $ proves false.

On the other hand, consider the case with $\zeta_\infty \ = \ 
    \psi^{y_2, -y_1, \theta}_{\om_0}$ for some value of $\theta$.
By \eqref{smallest} there exists a sequence $\theta_k$ such that 
\be \label{subsub}
\| \xi_k - \psi^{y_1, -y_2, \theta_k}_{\om_0} \|_Q \ \leq \ \f 2 3 
\ve_0.
\ee
Using elementary triangular identity, \eqref{choiceps} and
\eqref{subsub}, we 
obtain, for any $\theta \in [0, 2 \pi)$,
\be \nonumber
\| \xi_k -  \psi^{y_2, -y_1, \theta}_{\om_0} \|_Q \ \geq \ \|
\psi^{y_1, -y_2, \theta_k}_{\om_0} -  \psi^{y_2, -y_1, \theta}_{\om_0}
\|_Q
- \|  \psi^{y_1, -y_2, \theta_k}_{\om_0} - \xi_k \|_Q \ \geq \ \f
{\ve_{0}} 3.
\ee
This contradicts \eqref{poscontra},
% which implies $\xi_k \to
%\psi^{y_2, -y_1, \theta}_{\om_0}$ strongly in $Q$.
%This is absurd, 
so
the proof is complete.

\end{proof}


\section{Perspectives}
The interplay between nonlinearity and defect is, in our opinion, a
promising and worth developing field. In particular, 
already in simple models it makes highly non trivial
behaviour emerge. An enlightening example has been supplied by  means
of the $\delta^\prime$ defect, in which
the occurrence of a pitchfork bifurcation with symmetry
breaking has been proved for the family of nonlinear ground states.

Such results have to be considered as the first achievements of our
research project. Many non trivial variations on the theme could be given by studying the
 the entire family of one-dimensional defects (a four parameters family, see  
\cite{abd}) and thus investigate the effect of various self-adjoint
boundary conditions, in particular, of those that
give rise to {\em two} bound states. We expect that, in the nonlinear
problem, 
each of the two linear modes could be deformed into nonlinear modes
for any frequency greater than the energy of the corresponding linear
mode. Think, for instance, of a point interaction that, roughly
speaking, is the sum of a $\delta$ and a $\delta$-prime defect {\em at
  the same point}. It exhibits two bound states, one of which is even
(as the ground state for a Dirac's delta), while the other is odd (as
the ground state for a delta prime). A number of question then arises:
how do the corresponding nonlinear mode interact? Does it exist a
third family of stationary (possibly ground) states that does not
preserve any parity symmetry?

However, all these steps are only preliminary to the problem of
studying the evolution of a soliton that meets an impurity.

It remains completely open the problem of defining analogous models in
higher dimension. We recall that in dimension two and three, the only point interaction is the delta interaction, and in dimension higher than three there are no point perturbations of the laplacian. For instance, in the three dimensional case a bare
power nonlinearity seems to be too strong to be added to a Dirac's
delta potential; so a different type of nonlinearity with a moderated behaviour 
at infinity should be considered. Conversely, in space dimension two the na{\"{\i}}f
power nonlinearity could be not necessarily in conflict with the domain of a delta interaction, but up to now no rigorous result exists.

Another related topic is given by quantum graphs (\cite{kuchment1, 
kuchment2, ks}
for the relevant definitions and analysis in the linear case). Also in the relatively simple case of a NLS on a star graph, the richer structure provides a larger number of nonlinear
stationary states, for example two stationary states for a three edge star graph with a delta vertex, both attractive and repulsive, and the number increases with the number of edges (see \cite{[ACFN2]}). In this respect, besides the determination of the ground state, it is an open interesting problem the analysis of stability of excited states, here explicitly known. Nothing is known for the a star graph with more general vertex conditions, 
for example the boundary condition of $\delta^\prime$ type. 




\begin{thebibliography}{widest-label}
\bibitem{abgt} Adami R., Bardos C., Golse, F, Teta A.:
Towards a rigorous derivation of the cubic NLSE
in dimension one, {\em Asymp. An.}, {\bf 40}, 2, 93--108 (2004).
\bibitem{agt} Adami R., Golse, F, Teta A.:  Rigorous derivation of the
  cubic NLS in dimension one, {\it J. Stat. Phys}. {\bf 127} 1193-1220
  (2007).  
\bibitem{an1} Adami R., Noja D.: { Existence of dynamics for a 1-d NLS
  equation in dimension one}, J.Phys. A, {\bf 42} (2009), 49, 495302,
  19pp. 
\bibitem{[ACFN1]} Adami R., Cacciapuoti C., Finco D., Noja D.: Fast Solitons on Star Graphs,
Rev.Math.Phys, 23, 4 (2011)
\bibitem{[ACFN2]} Adami R., Cacciapuoti C., Finco D., Noja D.: Stationary states of NLS on star graphs, arXiv:1104.3839 (2011)
\bibitem{[AN2]} Adami R. Noja D., Stability: instability, bifurcation and symmetry breaking for the ground states  of a nonlinear Schr\"odinger equation with a $\delta^\prime$ interaction, {\it in preparation} (2011)
\bibitem{[ANS]} Adami R., Noja D., Sacchetti A.: On the mathematical description of the effective behaviour of one-dimensional Bose-Einstein condensates with defects, in {\it Bose-Einstein Condensates: Theory, Characteristics, and Current Research}, Nova Publishing, New York (2010)
\bibitem{as} Adami R., Sacchetti A.: {  The transition from diffusion to blow-up for a NLS equation in dimension one}, {\it J.Phys. A} {\bf 38} 8379-8392 (2005).
\bibitem{abd} Albeverio S., Brze\'zniak Z., Dabrowski L.: 
 Fundamental solutions of the Heat and Schr\"odinger 
Equations with point interaction,
 {\it Journal of Functional Analysis}, {\bf 130} (1995) 220-254.
\bibitem{[AGHH]} Albeverio S., Gesztesy F., Hoegh-Krohn R., Holden H.: 
{\it Solvable Models in Quantum Mechanics}. New York: Springer-Verlag 1988
\bibitem{[AK]} Albeverio S., Kurasov P.:   {\it Singular Perturbations of
Differential Operators}, Cambridge University Press 2000


\bibitem{ab}  Ammari Z., Breteaux S.: Propagation of chaos for many-boson systems in one dimension with 
a point pair-interaction, Arxiv preprint arXiv:0906.3047 (2009).

\bibitem{[BKP]} Brazhnyi V., Konotop V., Perez-Garc\'ia V.:
 Driving defect modes of Bose Einstein condensates in optical lattices, 
{\it Phys. Rev.Lett.} {\bf 96} 060403 (2006)
\bibitem{[BKP2]} Brazhnyi V., Konotop V., Perez-Garc\'ia V.: Defect modes of 
a Bose Einstein condensate in an optical lattice with a localized impurity,
  {\it Phys. Rev.A}, {\bf 74} 023614 (2006).


\bibitem{bl} Brezis H., Lieb E. H.: {\em A relation between
    pointwise 
convergence of functions and convergence of functionals}, 
Proc. Amer. Math. Soc. 88
486--490 (1983).

\bibitem{bcstv}
Burioni R., Cassi D., Sodano P., Trombettoni A., and Vezzani A.: \emph{Soliton
  propagation on chains with simple nonlocal defects}, Physica D \textbf{216}
  (2006), 71--76.


\bibitem{c} Cazenave T.: {\em Semilinear Schr\"odinger Equations},
  Courant Lecture Notes,  2003.

\bibitem{cs} Cheon T, Shigehara T.: Realizing discontinuous wave
  functions with renormalized short-range potentials, {\em
    Phys. Lett. A}, {\bf 243}, 22, 111-116 (1998).

\bibitem{dh}
Datchev K., Holmer J., \emph{Fast soliton scattering by attractive delta
  impurities}, Comm. Part. Diff. Eq. \textbf{34}, 1074--1113 (2009).




\bibitem{esy1} Erd\H{o}s L., Schlein B., Yau H-T.: Derivation of the
  Gross-Pitaevskii hierarchy for the dynamics of the Bose Einstein
  condensate. Comm.Pure Appl. Math. {\bf 59} 1659-1741, (2006). 
\bibitem{esy2} Erd\H{o}s L., Schlein B., Yau H-T.: Derivation of the
  cubic nonlinear Schr\"odinger equation from quantum dynamics of many
  body systems, Invent. Math. {\bf 167} 515-614, (2007). 


\bibitem{esy3} Erd\H{o}s L., Schlein B., Yau H-T.: Derivation of the
  Gross-Pitaevskii equation for the dynamics of the Bose Einstein
  condensate.,  Ann. Math., {\bf 172}, 1, 291--370 (2010). 



\bibitem{[EG]} Exner P., Grosse P.:
Some properties of the one-dimensional generalized point interactions (a torso),
mp-arc 99-390, math-ph/9910029  (1999). 
\bibitem{enz} Exner P. Neidhart H. Zagrebnov V.A., Potential approximations to $\delta^\prime$: an inverse Klauder phenomenon with norm-resolvent convergence, Commun. Math. Phys. 224, 593-612 (2001).
\bibitem{fibich}  Fibich G., Wang X. P.: Stability for solitary waves
  for nonlinear Schr\"odinger equations with inhomogenous
  nonlinearities, Physica D 175, 96--108 (2003).  
\bibitem{giannigianni} Fukuizumi R. and Jeanjean L., Stability of standing waves for a nonlinear Schr�Á�dinger equation with a repulsive Dirac delta potential, Discrete and Continuous Dynamical Systems (A), {\bf 21},  129-144 (2008).

\bibitem{reika} Fukuizumi R., Ohta M, Ozawa T.:  Nonlinear
  Schr\"odinger equation with a point defect, {\it
    Ann. I.H.Poincar\'e, AN}, {\bf 25}, 837-845 (2008). 

\bibitem{fukusacchetti} Fukuizumi R., Sacchetti A.: Bifurcation and stability for 
Nonlinear Schroedinger equations with double well potential in the 
semiclassical limit, preprint arXiv:1104.1511 (2011).

\bibitem{gow}
Goodman R. H., Holmes P. J., Weinstein M. I.: Strong NLS
soliton-defect interactions {\it Physica D}, {\bf 192},
215-248 (2004). 


\bibitem{gss1} Grillakis M. Shatah J. Strauss W.: Stability theory of
  solitary waves in the presence of symmetry, I J.Funct.Anal., {\bf
    74}, 160-197 (1987).  
\bibitem{gss2} Grillakis M. Shatah J. Strauss W.: Stability theory of
  solitary waves in the presence of symmetry, II J.Funct.Anal., {\bf
    94}, 308-348 (1990).   

\bibitem{hmz}  Holmer J., Marzuola J., Zworski M.: Fast soliton
  scattering by delta impurities, {\it Comm. Math. Phys}, {\bf 274},
  187-216 (2007).


\bibitem{jw} Jackson R.K, Weinstein M.I., Geometric analysis of
  bifurcation and symmetry breaking in a Gross-Pitaevskii equation,
  J. Stat. Phys., {\bf 116}, 881-905 (2004). 

\bibitem{kss} Kirkpatrick K., Staffilani G., Schlein B.:
Derivation of the two-dimensional nonlinear Schr���dinger equation from
many body quantum dynamics, {\em Am. J. of Math.}, {\bf 133}, 1,
91--130 (2011).

\bibitem{kevrekidis} Kirr E.W., Kevrekidis P.G., Pelinovsky D. E.: Symmetry-breaking bifurcation in the nonlinear
Schrodinger equation with symmetric potentials, preprint arXiv:1012.3921 (2010).

\bibitem{kirr} 
Kirr E.W., Kevrekidis R.G., Shlizerman E., Weinstein M.I., Symmetry-breaking bifurcation
in nonlinear Schr���odinger/Gross--Pitaevskii equations, {\em SIAM
  J. Math. Anal.}, {\bf 40} 566--604 (2008).





\bibitem{ks} Kostrykin V., Schrader R., {{K}irchhoff's rule for quantum wires}, J.
  Phys. A: Math. Gen. \textbf{32}, no.~4, 595--630, 1999.



\bibitem{kuchment1}
Kuchment P.: {Quantum graphs. {I}. {S}ome basic structures}, Waves Random
  Media \textbf{14}, no.~1, S107--S128 (2004).

\bibitem{kuchment2}
Kuchment P.: {Quantum graphs. {II}. {S}ome spectral properties of quantum and
  combinatorial graphs}, J. Phys. A: Math. Gen. \textbf{38}, no.~22,
  4887--4900 (2005).



















\bibitem{lacozza}  Le Coz S., Fukuizumi ., Fibich G., Ksherim B., Sivan Y.:
{\em Instability of bound states of a nonlinear Schr\"odinger
  equation with a Dirac potential},  Phys. D  237,  no. 8, 1103--1128 (2008).

\bibitem{l1} Lions, P.L.: The concentration-compactness principle in the calculus of variations.
The locally compact case. I {\em Ann. Inst. H. Poincar\'e Anal. Non Lin\'eaire},
{\bf 1}, 109--145 (1984).

\bibitem{l2} Lions, P.L.: The concentration-compactness principle in the calculus of variations.
The locally compact case. II {\em Ann. Inst. H. Poincar\'e Anal. Non Lin\'eaire},
{\bf 1}, 223--283 (1984).







\bibitem{tuoc} Pelinovsky D.E., Phan T.: Normal form for the
  symmetry-breaking bifurcation in the nonlinear Schr\"odinger
  equation, preprint arXiv:1101.5402v1 (2011).

\bibitem{p}
Perelman G., {A remark on soliton-potential interaction for nonlinear
  {S}chr\"{o}dinger equations}, Math. Res. Lett. \textbf{16}, no.~3,
  477--486 (2009).



\bibitem{stringari} Pitaevskii L., Stringari S.: {\em Bose-Einstein
    condensation}, Oxford University Press, 2003.

\bibitem{supersacchetti} Sacchetti A.: 
Universal Critical Power for Nonlinear Schr\"odinger Equations with a 
Symmetric Double Well Potential, {\em Phys. Rev. Lett.} {\bf 103}, 
194101 (2009).


\bibitem{vk} M. G. Vakhitov, A. A. Kolokolov: {\em Stationary solutions of the wave
equation in a medium with nonlinearity saturation}, Radiophys. Quantum Electron, 16,
783-789, 1973.
\bibitem{[W1]} Weinstein M., Nonlinear Schr\"odinger equations and sharp interpolation estimates 
Commun. Math. Phys., {\bf 87}  567-576, (1983) .
\bibitem{[W2]} Weinstein M., Modulational stability of ground states of nonlinear Schr\"odinger equations 
SIAM J. Math. Anal. Volume 16472--491  (1985) .
\bibitem{[W3]} Weinstein M., Lyapunov stability of ground states of nonlinear dispersive evolution equations 
Comm. Pure Appl. Math., Volume 39  51--68 (1986).
\end{thebibliography}

\end{document} 
  
---------------1202170308174
Content-Type: application/pdf; name="3halfsol-11feb12.pdf"
Content-Transfer-Encoding: base64
Content-Disposition: inline; filename="3halfsol-11feb12.pdf"
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---------------1202170308174
Content-Type: application/pdf; name="sesqui-11feb12.pdf"
Content-Transfer-Encoding: base64
Content-Disposition: inline; filename="sesqui-11feb12.pdf"
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---------------1202170308174--
